A PAVEMENT FEEDBACK DATA SYSTEM

by

.

Oren G. Strom W. Ronald Hudson James L. Brown

Research Report Number 123-12

A System Analysis of Pavement Design and Research Implementation

Research Project 1-8-69-123

conducted

in cooperation with the U. S. Department of Transportation Federal Highway Administration

by the

Highway Design Division Texas Highway Department

Texas Transportation Institute Texas A&M University

Center for Highway Research The University of Texas at Austin

May 1972

The contents of this report reflect the views of the authors, who are responsible for the facts and the accuracy of the data presented herein. The contents do not necessarily reflect the official views or policies of the Federal Highway Administration. This report does not constitute a standard, specification, or regulation. PREFACE

This report describes a conceptual feedback data system for flexible highway pavement design and is an extension of the research reported in Report 123-4, "Developing a Pavement Feedback Data System." It also represents a coordinated plan of action to activate the information subsystem identified in the initial project report 123-1.

Included herein is a discussion of the concepts of management information systems and their relationship to a pavement feedback data system (PFDS). Emphasis is placed on the collection and storage of carefully screened pavement data which meet strict essentiality criteria. The resulting data bank is a fundamental complement to the flexible pavement design system (FPS) now being pilot implemented in ten districts of the Texas Highway Department.

This is the twelfth in a series of reports emanating from the project entitled "A System Analysis of Pavement Design and Research Implementation." The project is sponsored by the Texas Highway Department in cooperation with the Federal Highway Administration, and proposes a systematic and comprehensive program to achieve improved pavement design methods.

Special appreciation is extended to Mr. Frank Yu for his technical advice and consultation in the area of systems analysis and computer programming. His efforts have been especially helpful in the resolution of numerous system automation problems and the development of workable file concepts for the proposed pavement data system.

The cooperation and assistance given by several Texas Highway Department and Center for Highway Research personnel are also sincerely appreciated. Messrs. Ben Barton, Tom Cartier, and Duval Jarl of the Texas Highway Department were particularly helpful in several phases of this research effort.

> Oren G. Strom W. Ronald Hudson James L. Brown

May 1972

iii

This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

LIST OF REPORTS

Report No. 123-1, "A Systems Approach Applied to Pavement Design and Research," by W. Ronald Hudson, B. Frank McCullough, F. H. Scrivner, and James L. Brown, describes a long-range comprehensive research program to develop a pavement systems analysis and presents a working systems model for the design of flexible pavements.

Report No. 123-2, "A Recommended Texas Highway Department Pavement Design System Users Manual," by James L. Brown, Larry J. Buttler, and Hugo E. Orellana, is a manual of instructions to Texas Highway Department personnel for obtaining and processing data for flexible pavement design system.

Report No. 123-3, "Characterization of the Swelling Clay Parameter Used in the Pavement Design System," by Arthur W. Witt, III, and B. Frank McCullough, describes the results of a study of the swelling clay parameter used in pavement design system.

Report No. 123-4, "Developing A Pavement Feedback Data System," by R. C. G. Haas, describes the initial planning and development of a pavement feedback data system.

Report No. 123-5, "A Systems Analysis of Rigid Pavement Design," by Ramesh K. Kher, W. R. Hudson, and B. F. McCullough, describes the development of a working systems model for the design of rigid pavements.

Report No. 123-6, "Calculation of the Elastic Moduli of a Two Layer Pavement System from Measured Surface Deflections," by F. H. Scrivner, C. H. Michalak, and W. M. Moore, describes a computer program which will serve as a subsystem of a future Flexible Pavement System founded on linear elastic theory.

Report No. 123-7, "Annual Report on Important 1970-71 Pavement Research Needs," by B. Frank McCullough, James L. Brown, W. Ronald Hudson, and F. H. Scrivner, describes a list of priority research items based on findings from use of the pavement design system.

Report No. 123-8, "A Sensitivity Analysis of Flexible Pavement System FPS2," by Ramesh K. Kher, B. Frank McCullough, and W. Ronald Hudson, describes the overall importance of this system, the relative importance of the variables of the system and recommendations for efficient use of the computer program.

Report No. 123-9, "Skid Resistance Considerations in the Flexible Pavement Design System," by David C. Steitle and B. Frank McCullough, describes skid resistance consideration in the Flexible Pavement System based on the testing of aggregates in the laboratory to predict field performance and presents a nomograph for the field engineer to use to eliminate aggregates which would not provide adequate skid resistance performance. Report No. 123-10, "Flexible Pavement System - Second Generation, Incorporating Fatigue and Stochastic Concepts," by Surendra Prakash Jain, B. Frank McCullough, and W. Ronald Hudson, describes the development of new structural design models for the design of flexible pavement which will replace the empirical relationship used at present in flexible pavement systems to simulate the transformation between the input variables and performance of a pavement.

Report No. 123-12, "A Pavement Feedback Data System," by Oren G. Strom, W. Ronald Hudson, and James L. Brown, defines a data system to acquire, store, and analyze performance feedback data from in-service flexible pavements.

ABSTRACT

The complex character of highway pavements coupled with ever-increasing traffic volumes and variability of climatic conditions have made it imperative that service life data be collected and analyzed to guide the design of new and reconstructed pavements. The road test approach inevitably omits some important aspects of performance and the mechanistic approach has thus far failed to yield the rational design models sought for so long.

Development of a system to collect, store and analyze carefully selected performance feedback data from full-scale, in-service pavements is proposed to overcome the inherent deficiencies of the mechanistic and road-test research techniques. This report is a case study example of such a data system for the State of Texas, and the selected performance factors are basically the inputs to a computer-based pavement design system known as Flexible Pavement System (FPS).

The most logical and efficient method of storing, retrieving, and analyzing vast quantities of data involves a modern electronic computer system programmed to perform typical file processing and management information system functions for the highway design and research engineer. Particular care must be taken to prevent intrusion of excess or irrelevant data into the system since this soon leads to system overloading and breakdown. The needs of the potential highway engineer users must govern system development to insure responsiveness.

A fundamental decision must be made regarding the record control key for the data system so that data is tied uniquely to that segment of highway pavement it describes. While many methods of record keying are in common use, there is a strong argument for using the existing Texas method for this system with appropriate, minor modifications.

There are several sources of feedback data and the methods of acquisition must be tailored accordingly. While some data needs can only be satisfied by initiation of new, statistically designed sampling and reporting procedures, there are already in existence a number of automated data files containing

vii

many factors specifically needed for the pavement design function. Explicit recognition of this fact allows design of interacting data files and elimination of duplication. A favorable environment for this kind of interaction has been provided for the Texas system by acquisition of the proprietary data handling system known as MARK IV. Files must be designed to readily accomodate data with important time and space dimensions. Analysis routines are necessary to give the file users meaningful data summaries from standard as well as custom-written mathematical models. Significant pay-off in the form of improved highway systems is anticipated from full implementation of this pavement feedback data system.

KEY WORDS: feedback data, computer, data acquisition, data analysis, data retrieval, design information, information retrieval, information systems, systems analysis, pavements, pavement management, research management.

SUMMARY

The data system defined in this report is a suggested plan for the discriminate acquisition and analysis of data needed to properly execute the Flexible Pavement System (FPS). The concepts of management information systems presented herein are applicable to any pavement design system, and the system defined may therefore serve as a model for eventual development of a comprehensive and integrated highway management system. Immediate application to the flexible pavement design system will allow checking of the design models and submodels, thus leading to development of new models/submodels as appropriate. Some basic management parameters are also included in the form of pavement performance factors and will serve to augment data now used by the District Engineer in his highway management program. A much more responsive data and information system is envisioned with no increase in personnel resources. This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

IMPLEMENTATION STATEMENT

The specific steps to practical application of these research findings are discussed in detail in Chapter 10 of this report. Two points merit emphasis: (1) implementation of a pavement feedback data system generally as outlined herein is absolutely imperative in an organized, methodical attack upon the pavement design problem, and (2) implementation of the proposed data system can be achieved without additional personnel by reordering priorities and reassigning resources to the acquisition and handling of only that data having specific functional uses. It may be anticipated that benefits will include improved pavement design procedures and a responsive management information system for the highway engineer. This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

TABLE OF CONTENTS

PREFACE	.ii
LIST OF REPORTS	v
ABSTRACT	<i>r</i> ii
SUMMARY	ix
IMPLEMENTATION STATEMENT	xi
CHAPTER 1. INTRODUCTION	
The Systems Approach	1 2
CHAPTER 2. DEFINITION OF A PAVEMENT FEEDBACK DATA SYSTEM	
Expert Observations	5 6 6 8
CHAPTER 3. EVALUATION OF GENERALIZED DATA MANAGEMENT SYSTEMS	
Hardware Constraints	11 11 13 13 14
CHAPTER 4. RECORD CONTROL KEY	
Definitions	15 15 16 17 18 19 25

•

CHAPTER 5. SELECTION OF ESSENTIAL FACTORS

	ntroduction and Background	27 29 32 37 38 39 40 45 46
CHAPT	R 6. EXISTING DATA FILES AND SOURCES	
	cope of Existing Files	47 47 52 54
CHAPT	R 7. DATA SAMPLING	
	ampling Concepts	55 55 57 57 58 61
CHAPT	R 8. FILE STRUCTURE AND DATA ANALYSIS verview of System and File Structure	63 65 69 72 72 77 77

CHAPTER 9. PFDS INFORMATION POTENTIAL

PFDS Equals	Information		•	•	٠	•	•	•	•	•	•	•	٠	•	•	•	٠	٠	85
Information	for District Manager	•	٠	•	•	•	•	•	•	•	•	•	٠	•	٠	•	٠	•	86

Custom-Written Analysis Programs

• •

. . .

79

80

. . .

. . .

Information for Administration	•		•	•	• •	•	•	•	•	•	•	•	•	88
Information for the Research Enginee	r	••	•	•	• •	•	•	•	•	•	•	•	•	92
CHAPTER 10. PFDS IMPLEMENTATION STRATEGY														
Administrative Actions and Decisions	•		•	•	•	••	•	•	•	•	•	•	•	95
Pilot Implementation	•	• •	•	•	• •	•	•	•	•	٠	•	•	•	96
Revision and Extended Implementation	•	• •	•	•	• •	•	•	•	٠	•	٠	•	•	97
Implementation Summary	•	• •	•	•	•	•	•	•	•	•	•	٠	•	98
CHAPTER 11. CONCLUSIONS AND RECOMMENDATI Conclusions	ons	 	•	•	• •	•••	•	•	•	•	•	•	•	101 101
REFERENCES	•	••	•	•	•	••	•	•	•	•	•	•	•	103
APPENDICES														
Appendix A. Abbreviations	_		_		_			_	_	-				113

	Appenaix A.		113
	Appendix B.	PFDS Specifications and Macro Comparison of GMDS	117
	Appendix C.	Traffic Log Coding Form and Sample File Output	131
	Appendix D.	Road Life (RL-1) Coding Form	137
	Appendix E.	Road Inventory (RI-2) Coding Form	141
THE	AUTHORS		143

.

CHAPTER 1. INTRODUCTION

Highway engineers have always sought a rational pavement design procedure. Some have said that classical mechanics would soon provide better design equations, while others have pursued the non-mechanistic, road test type approach. Meanwhile, the nation's automobile and truck traffic has grown at a furious rate, and vast sums of money have gone into the highway system. The pavement structure itself has been no small investment.

Unfortunately, much of the Interstate System has begun to show early distress and the General Accounting Office, the congressional watchdog on federal spending, has criticized the pavement designer and his works (Ref 80). W. N. Carey, Jr., in his opening remarks to the Highway Research Board (HRB) Workshop on Structural Design of Asphalt Concrete Pavement Systems, Austin, Texas, December 7, 1970, reminded his colleagues of these facts and admonished that "we better hurry to get some rational answers" (Ref 37). He went on to observe that this situation "is no longer a minor skirmish - an interesting intellectual exercise - it is a serious situation for all of us and for American transportation."

The Systems Approach

Although the road tests have provided answers to some questions, there remain the many unknowns that were beyond the limits of these experiments. The past five years has seen a great deal of emphasis placed upon a coordinated, systematic approach to pavement design. A cooperative research program involving the Texas Highway Department, the Texas Transportation Institute of Texas A&M University, and the Center for Highway Research of The University of Texas at Austin, has been underway since 1968 to provide some systematic improvements to the entire pavement design and management problem. Project 1-8-69-123, "A Systems Approach Applied to Pavement Design and Research," has yielded a number of findings and conclusions documented in a series of reports (Refs 28, 40, 45, 47, 56, 66, and 90).

Scope of Report

One of the fundamental objectives of the project has been to develop an information subsystem to yield research information, design information and feedback data (Ref 40). Figure 1 illustrates the basic characteristics of feedback information network envisioned in Project 123. Some preliminary planning guides for a data system were provided previously by R. Haas (Ref 28), and this report is a continuation to describe a Pavement Feedback Data System (PFDS) to be initially implemented for flexible pavements only. However, the concepts and principles of Management Information Systems (MIS) presented herein are equally applicable to rigid (or any) pavement system. Figure 2 depicts the various elements of this report and their relationship to the Pavement Design System (PDS) that has evolved from the research efforts of Project 123.

Chapter 2 presents the why of a PFDS, with special emphasis on the feedback loop from real in-service pavement systems instead of experimental sections.

Chapter 3 summarizes the state-of-the art in Generalized Data Management System (GDMS) software development, while the vital aspects of data control and coordination are discussed in Chapter 4.

Chapter 5 goes directly to the heart of the Flexible Pavement System (FPS) design method and sets forth those essential factors for which data should be collected, stored and analyzed in the feedback loop.

Chapter 6 is a summary of the existing data files of the Texas Highway Department, with main emphasis on those factors of direct application in the FPS design procedure.

Chapter 7 discusses sampling procedures to acquire reliable and representative data, and Chapter 8 presents conceptual and actual versions of file structures within the proposed PFDS.

Chapter 9 presents some potential user interactions with PFDS and illustrates some typical information output from the system.

Chapter 10 is a detailed implementation guide suggested for use by the Texas Highway Department.

Chapter 11 contains conclusions and specific recommendations for PFDSrelated research and administrative actions.





Fig 1. Feedback information flow (after Hudson et al, Ref 40).



Fig 2. Scope of a pavement feedback data system (PFDS).

CHOPTER 2. DEFINITION OF A PAVEMENT FEEDBACK DATA SYSTEM

Expert Observations

In a memorandum dated October 23, 1962, D. C. Greer, Texas State Highway Engineer, advised his Chief of Design of the increase in research funding to be expected as a result of the Federal Aid Highway Act of 1962 (Ref 27). In directing the initiation of an expanded research program to properly utilize the increased funding, he stated:

For many years, I have been convinced that the best research laboratory available to us is the Texas Highway System. ... It is possible that this is the time that we could take advantage of these funds that we will be required to expend to set up a research project to visit, inspect, and tabulate all information available on all concrete pavements built on the Texas Highway System in the past 25 years, together with the subsequent cost of maintenance and perpetuation. It is possible that such accumulation of data might then be handled through computers in such a manner as to give to us the tools whereby we might intelligently interpret the experiences of the past to guide us in subgrade and concrete pavement design in the future.

These words precisely describe the needed Pavement Feedback Data System (PFDS). Nearly eight years later, Karl Pister, Professor of Engineering Science at the University of California, stated the mechanist's viewpoint of a PFDS (Ref 61). In his paper, he referred to systematic and continuous observations of performance of full scale pavements and stated:

It is only through such a data acquisition program that any hope of pattern recognition will emerge to guide the formalization of operational rules leading to rational design. For example, without this, mathematical simulation of pavement systems, no matter how fascinating a game in itself, will remain precisely a game with very little pay-off to pavement systems.

Somewhat later in the same paper, Pister observed that rational but inadequate models of pavement behavior have been used successfully, primarily by allowing the engineer to use his judgment. He continued:

In other words, the engineer is a short circuit of the rational design process. Our attempts should be, it would seem, directed toward continued use of the engineer in this role but supplying him with the best possible data upon which to base his judgments, thereby minimizing the possibility of irrational short-circuits.

The preceding descriptions of a PFDS by a practicing engineer-manager and an engineering mechanist leave little more to say by way of definition and purpose. The title, Pavement Feedback Data System, is considered a logical and appropriate summarization. In a sentence, a PFDS is an automated system containing select feedback data from actual in-service highway pavements, to be used for research, design, and management functions.

The scope and purpose of PFDS can be illustrated in another way. The 1970-71 Pavement Research Needs Report (Ref 56) gave a detailed problem statement for the six most pressing research needs in the pavement design area, as indicated by prior Project 123 work of trial implementation and sensitivity studies. A close examination shows that every one of these needs has an explicit PFDS element. This is also implied in Fig 3 where PFDS is shown in relation to the Highway Research Information Service, an operational data source designed for a specific purpose. The importance of this latter statement is discussed more fully later in this chapter and in Chapter 9.

Management Information Systems

With caution, PFDS can also be viewed as an engineer's version of a Management Information System (MIS), the computer-age concept of providing the manager with all the information he needs to make the best decisions (Ref 31). But MIS's have not had a highly successful history so far. Robert V. Head, president of the Society for Management Information Systems, observes: "An MIS is something like the weather: Everyone talks about it but nobody does very much about it" (Ref 33). He goes on to point out that a manager's need for information is "ad hoc;" it cannot be predicted. J. Gosden, MITRE Corporation, agrees and points out that the manager deals with the exceptional cases (Ref 25). When exceptional situations begin to recur, they begin to receive standard action and pass from the realm of the manager to that of the foreman.

PFDS Design Objectives

PFDS is similar to a MIS in its <u>information</u> potential but different in its <u>data</u> concept (see Chapter 9). One of the fundamental design objectives has been careful selection of relevant data factors with special emphasis on elimination of irrelevant data. This is regarded as a principal objective



HRIS = Highway Research PFDS = Pavement Feedback Information Service Data System

Fig 3. Interaction of PFDS with the operating pavement system.

-

omitted in the design of most MIS's (Ref 16, pp 401-411). When emphasis is only on supplying relevant data, almost exclusive attention is given to generation, storage, and retrieval of information. Changing the emphasis to elimination of irrelevant data results in a great deal of redundant material in <u>relevant</u> documents being purged. This is appropriate and helps to prevent the manager from being swamped with more data than he can possibly use.

Another fundamental and closely related PFDS objective has been to structure the system to serve a specific group of users, specifically, the pavement engineers. In addition to limiting the scope of the data realm, it assures development of a system for which "customers" exist and need not be generated. The Wisconsin Department of Transportation designed and built their Highway Network Data and Information (HNDI) System largely independent of a specific group of users. When the system became operational, they found that no user really existed and that the next necessary step was thorough indoctrination of field personnel as to the scope, character, and possible users of HNDI (Ref 86).

PFDS Development Phases

In his preliminary work on PFDS (Ref 28), Haas outlined the steps or phases in design and development as shown in Fig 4. No further comment is necessary except to call attention to two observations he makes:

- (1) "Past experience has shown that it is very easy to underestimate the effort required to institute and maintain a comprehensive data system of this sort."
- (2) ".... The implementation must be done in stages."

Additional evidence has been found during this continuation of PFDS research to reemphasize these points (Refs 12, 24, 32, 35, 46, 49, 54, and 81). M. V. Jones of the MITRE Corporation provides an excellent discussion of computerization of government data systems in Ref 46. He estimates from 20 to 72 months for accomplishment of all tasks inherent in computerizing a data system depending on complexity, size, personnel, resources, etc. He also cautions that undue haste in the initial steps of the process can complicate and delay completion of later tasks. Roger A. MacGowan of the Department of Defense Computer Institute quotes an expert opinion in Ref 54 to the effect that system

Ş



Fig 4. Major phases in developing and applying a pavement performance evaluation scheme (after Haas, Ref 28).

designers have found the management information task to be a "far more complex, far more difficult process than they ever anticipated."

The several phases of PFDS are all addressed to a varying degree in the following chapters.

CHAPTER 3. EVALUATION OF GENERALIZED DATA MANAGEMENT SYSTEMS

Hardware Constraints

When the automation of a data system is being planned, two important considerations besides the data itself are:

- (1) hardware the computer, and
- (2) software the computer programs.

The hardware part of this problem is already solved in that the Texas Highway Department is already equipped with some of the most powerful and modern computer equipment available. In July 1971, THD installed the first of two new IBM System 370 Model 155 computers. These units replaced IBM 360 Model 50 units, so there never was any question about what brand of equipment would be handling the PFDS data files in an operational environment. By going to the new System 370, THD acquired four times the internal operating speed of their previous 360 units (Ref 18). This new computing power has already begun to show its effect and backlogged computing work has been eliminated.

Generalized Software Evaluation

Hardware is important for reasons other than computing speed. It literally locks a user in on certain computer programs and methods (software); and this is especially true in the area of generalized data management systems (GDMS), or in equivalent terms, general purpose software. This became an important consideration in PFDS research because it was logical to first seek a general purpose software package for data management rather than write one from scratch. There has been a good deal of effort made in the past 15 to 20 years to develop a general system that can be adapted to any situation.

John B. Glore of the MITRE Corporation (Ref 24) describes the potential advantages of a GDMS as

- (1) ease of use,
- (2) fast response,
- (3) economical use of computer equipment,

- (4) sufficient flexibility to configure and reconfigure solutions to a wide variety of problems, and
- (5) relatively little programming effort required.

Charles Kriebel, Carnegie-Mellon University, (Ref 50) states that there were about 50 such software systems commercially available in early 1969. Madill and Kuss of Simon-Fraser University (Ref 55) speculate that there are probably in excess of 100 existing software packages that perform data management functions. In any event, there is certainly a wide choice, both in vendors and price. A fairly recent summary (Ref 2) of file management systems by Altman, et al, reports that the commercial packages vary in cost from \$10,000 to over \$100,000.

With this kind of software marketplace, an effort was made to evaluate as many systems as possible for application to PFDS. The primary sources of information included Refs 7, 9, 21, 22, 24, 43, 44, 49, 58, 71, 79, 88, and 93. One of the difficulties encountered in this study was acquisition of objective information. Most available information is contained in vendor sales brochures and Jules Schwartz observes in Ref 54 that the seller's words have not been a notably accurate source for measuring the quality of software or its documentation. Objective evaluations by impartial authorities are limited.

Evaluation of 21 specific packages was undertaken by the Project 123 staff. All but five of these were promptly eliminated because of incompatibility with IBM hardware or the resident operating system, or because the package was no longer being maintained. The remaining five were (Appendix A)

- (1) DM-1,
- (2) COGENT III,
- (3) MARK IV,
- (4) GIS, and
- (5) NIPS.

Detailed specifications for PFDS software were prepared and grouped as mandatory, desirable, and optional. An attempt was then made to match the commercial package capability against the specifications. The result was that no system, as then defined, met all of the specifications for PFDS (see Appendix B). A tentative conclusion was made that a best all-around answer would be to write the software package patterned after a system prepared for the Montana Highway Department (Ref 93).

An Operational Decision

Concurrently, the THD Automation Division, D-19, was conducting an evaluation of generalized systems to be used for several departmental functions of personnel, equipment, and fiscal accounting. Regular coordination was maintained between D-19 and the Project 123 research staff to share information and findings. In February 1972, D-19 decided to purchase the MARK IV/260 file handling system marketed by Informatics, Inc.

This decision by D-19 is regarded as a very discriminate selection and immediately marked a turning point in PFDS planning. In the evaluation conducted by the Project 123 staff, MARK IV was noted as the only system that was in wide commercial use and acclaimed by the users as capable of everything claimed by the vendor. Even Glore acknowledged it as an apparently successful "limited system" in his discussion of GDMS shortcomings (Ref 24). However, its initial cost of \$40,000 was more than could be amortized at this time with PFDS alone. The use to be made of MARK IV in the entire Texas Highway Department is a completely different matter and adequately justifies such an expenditure.

MARK IV Application to PFDS

One of our first actions was acceptance of a D-19 invitation to look at MARK IV for possible application to PFDS. This first look revealed that the system has undergone constant improvement and now possesses capabilities not inherent in the earlier versions. One of the most important of these is an optional extra cost feature "Indexed Coordinated Files" that permits random access to the data base. Without this feature, the file can be processed sequentially only, a very slow and inefficient method of retrieving specific data items as needed. Random data access is considered an absolute must for PFDS and has always been a mandatory feature of the requisite software system. The Planning Survey Division (D-10) has indicated that this feature is essential for their file manipulations also.

Another extra cost feature of considerable value to PFDS is "Extended File Processing." This permits the simultaneous processing of nine files as compared with four in the basic MARK IV/260. This greatly improves processing time and efficiency and is also a valuable asset to D-10 operations. It is concluded the MARK IV with the Indexed Coordinated Files feature can do the PFDS job very adequately. The system possesses several characteristics that make its use for PFDS very promising:

- (1) Data fields are referred to by name.
- (2) File record structure may be changed with little effort to insert or delete data fields. This is a powerful asset in any research endeavor.
- (3) The coding forms are preprinted by the vendor and vastly simplify the use of the system (see Fig 28).
- (4) The files created by MARK IV are fully compatible with the existing operating system (OS) and are readily available for accessing by custom-written analysis routines.

A Master THD System?

In summary, it is important to emphasize that MARK IV offers many advantages to PFDS as well as to the Texas Highway Department. Perhaps the most important of these is that it creates a common environment for all THD files, thus allowing them to readily interact and interchange information. This is a vital asset in PFDS functions as will become evident in later discussion. It is also important to THD because it standardizes the terminology for use of all data files, regardless of content or principal user. This may represent an important first step to master THD files, with the concomitant benefit of true information exchange and elimination of duplication. CHAPTER 4. RECORD CONTROL KEY

Definitions

In the design of a computerized highway network data system, a method must be devised to uniquely identify data with a particular segment or point of roadway. The resulting device is a "record control key." If several data files are used to describe different highway system characteristics, it is important that the same record control key be used for all files in order to allow combined files processing and data analysis. This is a process called "correlation of data" and the control key is the common base of reference.

An analogy might be computer files on people in an organization. One file may contain personal data (height, weight, etc.) whereas two other files may contain previous employment records and current earnings data respectively. If we use the individuals' social security number as our record control key (in all files), we have uniquely identified the people in a consistent manner in our data files. We may therefore retrieve any combination of data pertaining to any particular person.

Methods of Location Control

The situation in regard to the highway system is similar but more complicated. Unlike people, our highway system is not a series of discrete units. Instead, it is a continuous ribbon of asphalt or concrete with constantly changing characteristics. Many different methods of location reference have been used to uniquely describe highway networks and William E. Blessing of the Federal Highway Administration discusses several in Ref 4. The principal methods are

(1) Route number and milepoint - Use of the federal or state highway number and a milepoint measured from some political or geographic feature such as state or county line. Field marking of the milepoint is done with uniformly or non-uniformly spaced mileposts. Data valid for only a point on a highway or for a given length of highway may be identified with this method.

- (2) Reference posts Unlike the milepost (or milepoint) method, reference posts each have a unique identification number without regard for the highway number. Central office records reflect the actual location of reference posts tied to route number, county, and milepoint from some starting point. Both length and point data may be tied to reference posts.
- (3) Route special feature log This method employs no field signing or posts. A straight line diagram or log is kept in a central office, showing the significant features encountered along the roadway. Highway intersections, bridges, county lines, etc. are the political, geographical, and man-made features with recorded milepoints that serve as the common base of reference.
- (4) Coordinates This method of position identification is exemplified by the international global method of latitude and longitude; a given latitude and longitude define a point on the earth's surface. On a smaller scale, state plane coordinates have frequently been used to locate specific points within a given state. This method necessitates several maps of the state along with a template or "Romer" to scale the coordinate locations (Ref 82). No field signing is used; data values are recorded in terms of north-south and eastwest coordinates.

The Texas System of Data Control

An additional method not discussed by Blessing is that used in Texas, the Control Section numbering system. Originated about 1935, the control section method was intended to provide a master reference system within which all subsequent physical and cost data could be tied to specific segments of the highway system. The system has been applied universally in Texas to all road systems over the years until today, all of the 70,000 miles of Texas highway are so identified. The following definitions apply (Ref 76):

- Control a length of roadway 50 to 100 miles long with well-defined geographic termini,
- Section a sub-unit or length of "control," typically 10 to 15 miles long with well-defined geographic termini, and
- Job Number a sequentially assigned number within the control section to identify special maintenance and/or construction work to be performed on a given segment of roadway at a given time. The job number may cover all or any fractional part of a control section, and jobs extending over more than one control section are assigned a separate job number within each control section.

This Texas system is therefore a three-level locational identifier having a time (or event) as well as space dimension. Although the controls and sections

were initially assigned in numerical sequence from west to east and from north to south, this uniform sequence has been interrupted by the irregular development of the state system. Today, there are 3,233 controls assigned and a total of about 8,000 control sections. This suggests that the "typical" control section is about 8-1/2 miles in length. It is important to emphasize that no portion of roadway has more than one control section number. However, frontage roads parallel to Interstate highways carry the same control section number as the Interstate.

Job numbers (the projects) vary in length and number per control section. The highest number assigned to date is 205; the typical job number length is 5 to 6 miles. The work they represent may vary in value from \$1,000 to over \$14 million. Special maintenance projects are usually lower dollar values whereas major construction and reconstruction are the high dollar project values. As might be expected, the high dollar value projects generally include major superstructures such as bridges, overpasses, interchanges, etc. It is also worthy of special note that each control section job number is carefully documented in a D-8 planning office set up specifically for this purpose. Pertinent data such as project description, limits, length, and cost are manually logged on standard Texas Highway Department forms.

Desirable Features

It should be understood that locational identifiers or record control keys for highway systems were not spawned by the computer, but they became imperative with the advent of computerized systems. In the development of such a data system, it is logical that a "best" record control key be sought. The attributes of such a key are

- (1) simple to use both in field and office,
- (2) absolutely unique for each section of roadway,
- (3) based on some logical progression or sequence,
- (4) relatively short symbol (code),
- (5) meaningful symbol (code) without extensive reference documents (map), and
- (6) really compensating for route changes, i.e., identity, location and length.

In PFDS research, a thorough review was made of all known existing methods and combinations. Some methods were strong in one area and weak in others; some were designed especially for item 6 above. Without resorting to any elegant techniques of evaluation, the various methods seemed to answer all desired attributes to about the same cumulative degree.

Because it is so different from the other methods, state plane coordinates deserve special comment. This method exploits the power of computer plotting techniques and possesses other characteristics so useful in many instances. It allows unique identification of a point rather than a cross section, with only two data values. In right-of-way and earthwork design situations, it allows computer processed plans, thus expediting the process and reducing manpower required. THD has a promising design system to accomplish just such work (Ref 3, pp 155-186). However, these many advantages are offset by the difficulty of using the method in the field. Other methods use the roadway centerline as one locational parameter, whereas state plane coordinates require two parameters of location, north-south and east-west coordinates. This requires elaborate control mechanisms not universally available and understood. Other problems recognized by John A. Vance, Toronto Transportation Systems Engineer, (Ref 82) are accuracy and data correlation. Thus, the coordinate method is considered a future improvement with much promise but not currently practical.

Blessing (Ref 4) concluded that the best method is a function of

- (1) installation and maintenance costs,
- (2) educational effort required, and
- (3) flexibility of the system.

For Texas, a fourth factor had to be considered, namely, convertibility from control section to whatever new method was selected. Closely related to the educational aspect, convertibility specifically means that data now embedded in control section can be successfully tied to the new locational identifiers.

Record Control Key Selection

During the evaluation process, a visit was made to the Wisconsin Department of Transportation (WISDOT) to discuss their newly established Highway

ĴŜ

Network Data and Information (HNDI) System. Oddly enough, WISDOT had been faced with the identical decision and had changed from log mile to the reference point method, backed up by state plane coordinates and "project numbers" (Ref 34). The back-up methods were intended to permit interaction of budget, accounting, and engineering design systems with the HNDI data base. Their experience was less than totally satisfactory. After considerable time and dollar expenditures, they were unable to assimilate the historical data on 12,000 miles of State Trunk Highway. This necessitated a resurvey of the system to obtain basic data. Furthermore, they have discovered that their reference point method occasionally yields some strange code designations, i.e., a "west" designation on a northbound lane of a divided highway (Ref 6). Their system employs the directional component in the key for a divided highway, and the anomaly occurs when the highway begins as a basic east-west roadway but runs north-south for appreciable distances.

Despite some deficiencies, it was concluded that the Texas control section method has far too many practical advantages to consider a completely new method. The most important advantages are:

- (1) The basic data on 70,000 miles of existing highway are all tied to control section and are excellently maintained.
- (2) The department procedures are now all based on control section.
- (3) The control section method works.

It is concluded that the existing control section method with some additional features should be used for PFDS.

Modifications Required

Some fundamental computer file concepts for PFDS will be presented in Chapter 8, but it is important here to recognize two basic computer system features that should influence development of a record key. First, the system feature called "Indexed Sequential Access Method" (ISAM) allows retrieval of data without sequentially searching the data tape until the desired values are encountered. To do this, the system must be given the unique (and exact) key for the data item (example: Social Security number precisely identifies a person). The second feature is related and is suggested by the word "precisely." If the precise key is not provided, the system (ISAM) will not function (except in specially programmed instances). Suppose, for example, we have stored data on a bridge in our data system using the simple key of highway number and milepoint:

IH 35 72.553

Later, we desire to retrieve certain data on this bridge and our milepoint measuring device yields a reading of 72.549. The system will not retrieve the data record unless a special search mechanism is written in the program. The MARK IV data management system does not have such a special search feature. Therefore, it appears wise to avoid such precise components in the record key.

As mentioned previously, the key should be as simple as possible. Those who have worked with these features in data systems will quickly agree that the human error potential increases with complexity of the code, and the computer is completely intolerant of errors.

The proposed PFDS record control key is a concatenation of the following identification components:

- (1) District 2 digits maximum,
- (2) Control 4 digits maximum,
- (3) Section 2 digits maximum,
- (4) Job Number 4 digits maximum,
- (5) Roadway 1 digit maximum, and
- (6) Lane 1 digit maximum.

The district component (field) is suggested as an aid to rapid retrieval, especially for summary type reporting. The remaining fields are shown in Fig 5. The crosshatched area of the westbound lanes depicts an overlay, job number 100 in one control section and 50 in the other. We are concerned here only with control section 2374-2. Note that the roadway and lane codes are simple numerical designations left to right, looking in the direction of increasing milepoint. Paved shoulders are treated as separate lanes and numbered accordingly.

Before proceeding further with the example, two points in regard to job number should be emphasized. First, the permitted code is seen to be four digits whereas the highest number assigned to date is only three digits (205). The last digit is a decimal place, to indicate sub-units of a job number. For example, a job number code 2051 means 205.1, the first of two or more subdivisions of job number. The provision of this breakdown of job number is made



Fig 5. Illustration of record control key components.
necessary to properly identify changes in design, construction, or performance characteristics within a job number. In some cases, great variation in subgrade materials may necessitate a change in design within the job and such discontinuity must be recorded. In the vast majority of cases, the last place of the job number will be zero, indicating consistent design (or construction, etc.) for the entire job number length. In our example in Fig 5, the numbers 100 and 50 mean that job numbers 10 and 5 are consistent designs for the entire length of the project (job number).

The second point is that the termini for job numbers are indicated by milepoints included in the data part of the record. These essential locational features are parts of a space and time component of the data record as explained in Chapter 8.

The illustration here depicts the most complex situation probably to be encountered in reasonable practice. If a future need is shown for collecting and storing data for elaborate geometric situations such as interchanges, etc., a refined keying method may be necessary.

The serviceability-performance diagram for our example in Fig 5 is shown in Fig 6. This is for the westbound lanes only. Complete reconstruction was done as job number 50 in 1960 and brought the serviceability index (SI) up to 4.2. In 1965, a seal coat was applied as job number 70. Performance dropped to minimum acceptable in 1967 and an overlay was placed as job number 100, the one shown in Fig 5. Another special maintenance job and an overlay are to be accomplished before complete reconstruction in 1981 as job number 220. The missing job numbers (60, 80, 90, etc.) are work projects of no consequence to the pavement engineer, such as painting centerline and edge stripes and repair of bridge railings.

The collated data records for this pavement segment are illustrated in Fig 7. The district is number 2 in this case. It is seen that each record contains data on a unique segment of roadway at a unique time (occurrence of event).

Some meditation on this system of record control will generate the question as to how succeeding data entries are made. For example, if one of the data fields is serviceability index (SI), we may wish to enter 10 such readings each year for 20 years on our particular road segment. These additional data values will be contained in a supplementary file with a record key basically



Fig 6. Serviceability-performance diagram for westbound lanes of control section 2374-2 (Fig 5).

1.2 C2



Fig 7. Collated data records for highway segment illustrated in Figs 5 and 6.

the same as the master record. Explanation of this situation is provided in Chapter 8; it is sufficient here to recognize that no radical changes in method are needed to meet such situations.

Although this proposed record control key is believed to be sufficiently definitive for PFDS at this point, it should be observed that one additional, one-digit, field will specify wheel path also. The concept used in roadway and lane designations is equally applicable with no difficulty. It is also important to mention in passing that the roadway designation coding proposed here is analogous to a data field called "Travel Class" in the D-10 Road Life (RL-1) file (see Chapter 6).

It is especially important to recognize an important benefit to be accrued with this proposed record control key. As discussed in Chapter 6, some especially applicable and valuable data files now being maintained by the Planning Survey (D-10) Division are keyed with control section and milepoint, and in some cases, with job number as an added field. This means that conditions are present for complete compatibility of files, i.e., PFDS with existing D-10 files.

<u>Conclusions</u>

As a concluding thought, it has been suggested that certain deficiencies in the actual use of the control section and milepoints (or mileposts) have made the method less than completely effective. For example, there are reported instances where mileposts have not been changed from the old south to north increasing sequence. Others have reported that mileposts are missing, thus making field locational identifications difficult or impossible. It is axiomatic that no system or method, however good in concept, can work if not fully implemented. Certainly, the system most easily understood has a better chance of working than a new method which must be learned. It then becomes a matter of making existing methods work in accordance with established directives.

This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

CHAPTER 5. SELECTION OF PFDS ESSENTIAL FACTORS

Introduction and Background

The preceding discussion of management information systems (MIS) has stressed simplicity and the importance of collecting and storing only relevant data. A tenacious effort has been made to abide by these principles in the PFDS research. Like business executives, engineers have often been drowned in information, and the result may be manifested somewhat in the lingering reluctance highway engineers have for using computer-based information systems. Stacks of unused computer output have all too frequently been the visible evidence of a working MIS. The HRB Workshop on Structural Design of Asphalt Concrete Pavement Systems (Ref 37) recognized this situation as "data pollution" and concluded that it is a primary factor in abuse and misuse of data systems. Thus, this stage of PFDS development is perhaps the most important of all. The term "essential" has therefore been employed to convey the idea that only those factors absolutely necessary for an operating feedback data system are to be included. The decisions to exclude are difficult ones.

Relation to Other Functions

Before proceeding further, explicit recognition of the pavement design engineer's part in the overall highway system may serve to abate fears that disproportionate influence upon management is sought. While the serviceability of the highway system, as discussed herein, is the riding quality of the surface as expressed by serviceability index (SI), it is fully acknowledged that there is much more to the problem than the pavement system (or subsystem). Just as there are many types of highway maintenance efforts that have nothing to do with the pavement, so are there many considerations other than pavement structural adequacy that determine when a highway must be reconstructed. Figure 8 illustrates some elements of this puzzle. The highway system managers use all the pieces to arrive at a decision. It is not an objective herein to suggest the relative sizes of these pieces of the puzzle. All efforts are



bent to furnishing the best possible pavement piece, one that is derived from logical analysis of the synergy of the pavement system components.

Early in this project, Haas (Ref 29) provided some introductory concepts relative to a pavement feedback data system that gave the problem an overall dimension. He showed that pavement performance, described by serviceability index (SI) was a function of at least three factors:

- (1) Climate
- (2) Traffic
- (3) Pavement

Each of these factors is of course composed of several components or subfactors, some of which have been defined and some that probably have not yet been identified. Conceptually, the pavement performance situation can be represented as shown in Fig 9. The solid lines enclose factors and subfactors already recognized and identified as important while the dotted lines enclose factors not yet identified or adequately described.

Within the framework of Project 123, models and submodels have been defined to express, as well as the present state-of-the-art will allow, the activity within each pavement performance component. These models are derived from the vast reservoir of past research findings and adapted to conditions in Texas. The basic underlying foundation is the AASHO road test conducted in 1959-62 (Ref 8). This was followed by work at Texas Transportation Institute, the Texas Highway Department, and the Center for Highway Research (Refs 40 and 68). Superimposing the results of this work on the conceptual representation of the problem in Fig 9 we can now give names to the models, factors, and subfactors that make up our pavement design problem (see Fig 10).

The broken line boxes and vectors represent those models and subfactors, respectively, which are potentially present but not yet defined or even identified within the present context of the pavement design or performance problem. At this point in time, it can only be said that such factors are not a part of any current model. Continuing research and analysis will undoubtedly provide the identities in the future.

Future Additions

This PFDS research has precipitated a philosophy about pavement design in general and the PFDS in particular. Specifically and with some carefully



Fig 9. Components of pavement performance.

Legend:



Fig 10. Flexible pavement system (FPS) design components.

selected exceptions, the variables or factors identified in the described process of model analysis are the only ones for which data will be collected and stored at the start. As pointed out in Chapter 2, the initial goal of PFDS is to provide fundamental research and management tools, later maturing into a design and more complete management complement. However, the first data inserted into PFDS will be used primarily for select management functions and to check the various design models. During this model checking process, it is important that results of other, even though directly related, researchin-progress not be inserted into the operational design methods until a distinct need for such is indicated by performance data. Even though some theoretical breakthroughs may be apparently concluded during this period of performance data collection and analysis, implementation should be effected only upon demonstration of an omission or change in performance prediction that call for a rational revision to the applicable model(s). Such a policy should help to prevent premature changes in design procedures and assure that modifications to models are made in the proper priority.

Some research in progress can be used to illustrate this point. M. Y. Shahin, a graduate student on Project 123 at The University of Texas, has developed distress prediction models for asphalt pavements that may soon allow refinement of design techniques to limit temperature cracking (Ref 70). This research is an attempt to quantify the solution to a problem which has plagued engineers for years. However theoretically sound it may now appear, no attempt should be made to introduce the models into operational design systems until they are verified by performance data. In many respects, operational design systems are also experimental processes and introduction of new factors and models before performance feedback is acquired simply enlarges the inference space and introduces new and unknown interactions. In other words, while the pavement engineer may have a reliable estimate of the direction in which he is headed or diverted, he cannot be sure of his starting point. If PFDS is allowed to function in conjunction with the operational pilot design system, a reliable origin can be established and the engineer may proceed more accurately toward his goal.

Current Design System

The current operational pilot design procedure for flexible pavements used by THD is designated FPS-11 and uses the inputs shown in Table 1. The

TABLE	1.	VARIABLE	INP	UTS	TO	FPS	-11	DESIG	N
		PROGRAM A	AND	SOUR	CE	OF	VALU	ÆS	

	Variables	How Values Obtained				
		Assign	Access	Compute	Measure	
Bas	ic Design Criteria					
1.	Length of analysis period	Х				
2.	Minimum time to first overlay	Х				
3.	Minimum time between overlays	Х				
4.	Minimum serviceability index	Х				
5.	Design confidence level	Х				
6.	Interest rate	х				
Pro	gram Controls and Constraints					
1.	Maximum funds available per square yard for initial construction	х				
2.	Maximum total thickness of initial construction	х				
3.	Maximum total thickness of all overlays	х				
Tra	ffic Data					
1.	ADT at beginning of analysis period		х			
2.	ADT at end of analysis period		Х			
3.	One-direction cumulative 18-kip single axles during analysis period		х	х		
4.	Average approach speed to overlay zone	х			x ⁽¹⁾	
5.	Average speed through overlay zone – overlay direction	X			x ⁽¹⁾	
6.	Average speed through overlay zone - non-overlay direction	х			x ⁽¹⁾	
7.	Percent ADT arriving each hour of construction		х			
8.	Percent trucks in ADT		Х			
	Bas 1. 2. 3. 4. 5. 6. Pro 1. 2. 3. Tra 1. 2. 3. 4. 5. 6. 7. 8.	VariablesBasic Design Criteria1. Length of analysis period2. Minimum time to first overlay3. Minimum time between overlays4. Minimum serviceability index5. Design confidence level6. Interest rateProgram Controls and Constraints1. Maximum funds available per square yard for initial construction2. Maximum total thickness of initial construction3. Maximum total thickness of all overlaysTraffic Data1. ADT at beginning of analysis period2. ADT at end of analysis period3. One-direction cumulative 18-kip single axles during analysis period4. Average approach speed to overlay zone5. Average speed through overlay zone - overlay direction6. Average speed through overlay zone - non-overlay direction7. Percent ADT arriving each hour of construction8. Percent trucks in ADT	VariablesAssignBasic Design CriteriaX1. Length of analysis periodX2. Minimum time to first overlayX3. Minimum time between overlaysX4. Minimum serviceability indexX5. Design confidence levelX6. Interest rateXProgram Controls and ConstraintsX1. Maximum funds available per square yard for initial constructionX2. Maximum total thickness of all overlaysXTraffic DataX1. ADT at beginning of analysis periodX3. One-direction cumulative 18-kip single axles during analysis periodX4. Average approach speed to overlay zoneX5. Average speed through overlay zone - overlay directionX6. Average speed through overlay zone - non-overlay directionX7. Percent ADT arriving each hour of constructionX8. Percent trucks in ADTX	VariablesHow Value Assign AccessBasic Design Criteria1.1. Length of analysis periodX2. Minimum time to first overlayX3. Minimum time between overlaysX4. Minimum serviceability indexX5. Design confidence levelX6. Interest rateX7 Program Controls and ConstraintsX1. Maximum funds available per square yard for initial constructionX2. Maximum total thickness of initial constructionX3. Maximum total thickness of all overlaysXTraffic DataX1. ADT at beginning of analysis periodX3. One-direction cumulative 18-kip single axles during analysis periodX4. Average approach speed to overlay zoneX5. Average speed through overlay zone - overlay directionX6. Average speed through overlay zone - non-overlay directionX7. Percent ADT arriving each hour of constructionX8. Percent trucks in ADTX	VariablesHow Values Obtain Assign Access ComputeBasic Design CriteriaX1. Length of analysis periodX2. Minimum time to first overlayX3. Minimum time between overlaysX4. Minimum serviceability indexX5. Design confidence levelX6. Interest rateX7 Program Controls and ConstraintsX1. Maximum funds available per square yard for initial constructionX2. Maximum total thickness of initial constructionX3. Maximum total thickness of all overlaysXTraffic DataX1. ADT at beginning of analysis periodX3. One-direction cumulative 18-kip single axles during analysis periodX4. Average approach speed to overlay zoneX5. Average speed through overlay zone - one-overlay directionX7. Percent ADT arriving each hour of constructionX8. Percent trucks in ADTX	

(1) Elements of a potential special research project.

TABLE 1. (Continued)

Variables			How Values Obtained				
			Assign	Access	Compute	Measure	
D.	Env	ironment and Subgrade					
	1.	District temperature constant	Х		x ⁽²⁾		
	2.	Swelling probability - clay subgrade				X·	
	3.	Potential vertical rise – clay subgrade	x ⁽³⁾		х	Х	
	4.	Swelling rate constant - clay subgrade	x ⁽³⁾			х	
	5.	Subgrade stiffness coefficient			Х		
E.	Con	struction and Maintenance Data					
	1.	Initial serviceability index	x ⁽³⁾			х	
	2.	Serviceability index after overlaying	x ⁽³⁾			Х	
	3.	Minimum overl a y thickness	Х				
	4.	Overlay construction time	х				
	5.	Asphalt concrete compacted density	х				
	6.	Asphalt concrete production rate	Х			Х	
	7.	Width of each lane	x ⁽³⁾	x ⁽⁴⁾			
	8.	First year cost of routine main- tenance	x ⁽³⁾		x	х	
	9.	Incremental increase in maintenance cost per year	x ⁽³⁾	х	х		
F.	Det	our Design for Overlays					
	1.	Detour model used during overlay	х				
	2.	Total number of lanes of the facility		Х		x ⁽¹⁾	

 $^{(1)}$ Elements of a potential special research project.

(2) Adjusted values may be computed from maximum and minimum daily temperatures for the specific locality.

(3) For design purposes.

(4) In Road Life file (RL-1) as plan quantity; see Chapter 6.

TABLE 1. (Continued)

Variables			How Values Obtained			
			Assign	Access	Compute	Measure
	3.	Number of lanes open in the overlay direction	X			x ⁽¹⁾
	4.	Number of lanes open in the non-overlay direction	х			x ⁽¹⁾
	5.	Distance traffic is slowed - overlay direction	х			x ⁽¹⁾
	6.	Distance traffic is slowed - non-overlay direction	Х			x ⁽¹⁾
	7.	Detour distance around the overlay zone	X			x ⁽¹⁾
G.	Exi	sting Pavement and Proposed ACP				
	1.	SCI of existing pavement				х
	2.	Standard deviation of SCI			Х	
	3.	Composite thickness of existing pavement		x		
	4.	In-place cost (compacted) C.Y. of proposed ACP	X			
	5.	Proposed ACP's salvage value as percent of original cost	X			
	6.	In-place value of existing pavement-compacted C.Y.	Х			
	7.	Existing pavement's salvage value as percent of present value	X			
	8.	Level-up required for the first overlay	x			
	9.	Number of years existing pavement has been open to traffic		x		
Н.	Pav (Al con	ring Material Information I for each material to be sidered in design.)				
	1.	Layer designation	X	x ⁽⁴⁾		Х

(1) Elements of a potential special research project.

(4) In Road Life file (RL-1) as plan quantity; see Chapter 6.

35

(Continued)

	Variables	How Values Obtained				
		Assign	Access	Compute	Measure	
2.	Name of material	Х	x ⁽⁴⁾		х	
3.	In-place cost - compacted C.Y.	Х	Х		Х	
4.	Stiffness coefficient	Х			Х	
5.	Minimum allowable thickness	Х				
6.	Maximum allowable thickness	Х				
7.	Material salvage value as percent of original cost	х				

TABLE 1. (Continued)

(4) In Road Life file (RL-1) as plan quantity; see Chapter 6.

background and explanation of this computerized design system are contained in a report entitled "A Systems Approach Applied to Pavement Design and Research" (Ref 40). Also shown is the probable source of the data values. The following definitions regarding how values are obtained apply:

- "Assign" means that values are selected by the designer based on engineering judgment or as reasonable program constraints to prevent unreasonable design results.
- "Access" means that values are obtained from existing data sources such as the THD D-10, D-8, or D-18 data files.
- "Compute" means that values are obtained from computation upon raw data either accessed or measured.
- "Measure" means that field sampling of data values is or will be necessary.

It will be noted on the second page of the table that certain variables (marked by note 2) are shown to have values "assigned" as well as "measured." This situation occurs whenever insufficient performance (measured) data is available, or whenever two different times in the design, construction and performance process are inherent in the particular variable values. An example of this is initial serviceability index, 1 in group E. A design value, which represents the mean value for new flexible pavements in Texas, will be assigned. When the pavement is constructed, the actual serviceability index will be measured and compared to the design value. The resulting difference is a measure of the variability in construction practices and techniques.

Future Research Projects

Under groups C and F of Table 1, nine variables are designated as potential special research items for measurement (note 1). All of these items are used to compute the "user-costs" incurred during overlay projects at one or more times in the future use of the particular highway system (Ref 67). The user-cost models tend to oversimplify a very complex situation and this whole matter could easily use a separate research effort. Instead of collecting data on every detour used in practice, a well-designed experiment will probably yield more discernible results in a more efficient manner.

The note 4 refers to the Road Life file (RL-1) of D-10. This file will be discussed in detail in Chapter 6; however, at this point, it is sufficient to recognize that the indicated variables are currently included in an existing file and the data should serve adequately for future reliability and stochastic research within Project 123.

Essential Factors for PFDS

The basis for variables to be included in PFDS must necessarily come from FPS-11; input variables to that program are the basic essential factors for PFDS. However, these factors have both a time and space dimension that must also be sampled. All factors will exhibit variability as we progress down a roadway; for example, as-built surface thickness will vary around some mean value, possibly the design or plan quantity. Other factors will vary with time; for example, serviceability index will tend to decline with time from some high initial value. The handling of this situation will be evident in the discussion of the PFDS files later in the chapter. The acquisition of time-dependent data values is also discussed in Chapters 4 and 8.

Specific recognition of the time and space dimensions of the factors, and the formation of a plan to sample accordingly, provides the raw data needed to further the probabilistic and reliability concepts now being actively pursued in the Flexible Pavement System (FPS) (Refs 13 and 15). Darter (Ref 14) considers three types of variability in his reliability concepts

- (1) variability within a project such as pavement thickness, material strength, etc.;
- (2) variability between design (plan) quantities and as-built values; and
- (3) variability due to lack-of-fit of design models in FPS-11.

Recognition of these sources of variation serves as an important second development parameter for PFDS, specifically, the need to be adequately responsive to the active research in reliability and stochastic applications in FPS.

Just as important in any sampling plan is the determination of methods or techniques for taking data measurements. The methods must be standardized in so far as possible so that direct comparisons are valid. For example, the serviceability index of all highways must be determined by devices all calibrated to a common reference device. This is done in the Texas method (see Ref 85). Another example is the Texas Triaxial Class, a material property that is fundamental to a standard design procedure used in Texas (Ref 74). Because the cost of this test is high, the triaxial class is often determined from one of several possible "shortcut" techniques. Therefore, the record must specify the method which was actually used. This is discussed in more detail in Chapter 7.

Undefined Influences Upon Performance

Before proceeding to definition of the proposed PFDS files and essential factors, it will be helpful to examine again the typical serviceability/performance diagram shown in Fig 6 of Chapter 4. Each of the abrupt increases in the performance index values represents an event that should be recorded to permit complete subsequent analysis of the pavement system. Seal coats and overlays are certainly two significant events to record. Beyond those two, however, problems are promptly faced. What maintenance work affects the serviceability/performance life to an extent that it should be recorded for analysis? How much cracking must exist before sealing becomes significant? Cracking influence upon <u>present</u> serviceability index (SI) was quantified in the AASHO Road Test equations (Ref 8) but no satisfactory method of relating distress to performance (failure) has been derived. When sufficient data are collected, researchers will be in a position to ascertain this relationship more readily. The potentials of PFDS are described in Chapter 9.

The same thing can be said for other types of distress upon which important maintenance is performed. The Texas Highway Department Maintenance Manual (Ref 75) recognizes five types of asphaltic concrete distress and prescribes repair and prevention methods. Most of these maintenance and repair efforts will be specifically identified and pursued as individual job numbers in the regular THD accounting system, as discussed in Chapter 4, and the basic parameters of cost and general description can be easily included in PFDS. However, a distress survey something like that now being proposed in a new research study (Ref 57) is needed. The findings should be implemented as a future revision to the PFDS files and essential factors. In summary, it is fully acknowledged that maintenance efforts are very important in pavement performance and life, but until the significant factors can be explicitly defined in the FPS context, detailed distress and maintenance data collection and storage are premature. During the interim, maintenance costs can serve as a substitute parameter to measure distress.

The PFDS Files and Factors

Definition of the PFDS files and essential factors is now appropriate. A harsh interpretation of the PFDS objectives suggests that only the inputs to the FPS-11 design method should be included. However, one exception may be in order. Since Triaxial Class is a fundamental parameter in an existing manual design method (Ref 74), its inclusion in PFDS may be warranted to satisfy essential near-term design needs until FPS becomes a universal THD method. Thus it has been included in the proposed structural file.

Table 2 provides a complete listing of proposed files and essential factors. Six logical groupings of factors have been identified, each group to correspond to a master file in PFDS. Each master file may have one or more "trailer" files to provide additional detailed data as necessary. This concept is discussed in detail in Chapter 8. In addition, each file will contain certain time and locational data entries to positively identify the data by sample phase (design, as-built, or subsequent), date, and specific highway location. These features are also discussed in Chapter 8.

The following explanation of file features in Table 2 is appropriate:

- Structural File: contains the factors that uniquely describe the pavement structure. The file will have variable length records depending on number of layers and future overlays (events). The following additional comments are applicable:
 - (a) The sample phase for this file will always be design or as-built, i.e., the value used in design or the as-built feature/value.
 - (b) "Design comments" allow the insertion of the reason(s) for a design strategy or a subsequent event (overlay or seal coat).
- (2) Environment File: contains the essential environmental factors used by FPS-11 to estimate effects upon performance. The first is readily available as a computed constant at the present time. However, its value is derived from minimum and maximum daily temperatures for the individual district headquarters. More precise values can be readily computed for the specific district area where a given project is located by acquiring the National Weather Service (NWS) records for the nearest NWS temperature station. There are some 1,000 such stations in Texas and a single magnetic tape containing these data for 22 years (1948-1970) can be obtained for a nominal fee from the Austin office of the Texas Water Development Board. Once computed for several sub-areas within a district, such a temperature constant based on 22 years of data should require little if any future adjustment. This could be a one-time effort and eliminate all future temperature data collection (accessing).

TABLE	2.	PFDS	FILES	AND	ESSENTIAL	FACTORS
-------	----	------	-------	-----	-----------	---------

			aian	Sample P	hase	Data Source
		<u>U</u>	esign	<u>AS-DUIIL</u>	Subsequent	II Other Inan PRDS
Α.	Str lay	uctural (Fields l- er including subgr	5 for ade.)	each		
	1.	Layer designation and code	x	Х		Final plan quantities in D-10 Road Life (RL-1) file.
	2.	Material name	Х	Х		17 17 11
	3.	Layer thickness	Х	Х		11 Fr 11
	4.	Stiffness coef- ficient	х	Х	Х	
	*5.	Triaxial class	Х			
	6.	Design comments	Х			Coded comments in D-10 RL-1 file.
	7.	Lane width	Х	Х		Final plan values in D-10
	(Fi un	elds 8-11 for each til complete recon	subse	equent ev tion.)	ent	RL-1 file; inventory values in Road Inventory (RI-2) file.
	8.	Event (type work)	Х	Х		Plan values in RL-1 file.
	9.	Date	Х	Х		Same as for factor 6.
	10.	Layer thickness	Х	Х		Same as for factor 8.
	11.	Design comments	Х			Same as for factor 6.
Β.	Env	ironment				
	1.	District (local temperature con- stant	Х		Х	National Weather Service
	2.	Subgrade swelling probability	x	x ⁽¹⁾	Х	
	3.	Subgrade potentia vertical rise	.1 X	x ⁽¹⁾	Х	
	4.	Subgrade swelling rate constant	X	x ⁽¹⁾	Х	

* Exclude from surface layer data field.

(1) Opinion of construction engineer.

				Sample Phase		Data	Source	
			Design	<u>As-built</u>	Subsequent	If Other	Than PFDS	
С.	Per	fo rma nce						
	1.	Surf ac e Curvatur Index	e X	Х	Х			
	2.	Service a bility Index	Х	Х	Х			
	3.	Minimum Service- ability Index	X		Х			
	4.	Design Confidenc level	e X		Х			
D.	Tra	ffic						
	1.	ADT-initial	Х	Х	Х	D-10 Traffic	Log	
	2.	ADT-final	Х	Х	Х	D-10 Traffic	Log	
	3.	18 KSA Equivaler	it X	Х	Х	D-10 Traffic	Log	
E.	Cos	ts						
·	1.	Initial Construc tion \$/sy	:- Х	Х				
	2.	Overl a y Construc tion \$/sy	:- Х	Х				
	3.	Spe cia l Maintena \$/sy	nce X	Х				
F.	Con	straints	(De	sign v a lu	es only)			
	1.	County						
	2.	Highway						
	3.	IPE (Investigati	on Plan	nning Exp	ense Number)		
	4.	Length of analys	sis per:	iod				
	5.	Minimum time to	first	overlay				
	6.	Minimum time bet	ween o	verl a ys				
	7.	Interest rate						
	8.	Problem type						
	9.	Maximum funds av	ailable	e per S.Y	. for initi	al constructi	on	

	Sample Phase Data Source Design As-built Subsequent If Other Than PFDS
10	Maximum total thickness of initial construction
11	Maximum total thickness of all overlags
12	Average approach speed to overlay some
13	Average speed through overlay zone - overlay direction
14	Average speed through overlay zone - overlay direction
14.	Percent ADT arriving each hour of construction
16	Percent trucks in ADT
17	Minimum evenley, thickness
10	Prophy construction time
10	Apphalt concrete compacted density
20	Asphalt concrete compacted density
20.	Right weer east of routing reintergraph
21. 22	Incremental increases in maintenance
22. Do	Determined in maintenance cost/year
23.	Detour model
24.	lotal number of lanes
25.	Number lanes open in overlay direction
26.	Number lanes open in non-overlay direction
27.	Distance traffic is slowed - overlay direction
28.	Distance traffic is slowed - non-overlay direction
29.	Detour distance around overlay zone
*30.	Composite thickness of existing pavement
*31.	In-place cost/compacted C.Y. of proposed ACP
*32 。	Proposed ACP's salvage value as percent of original cost
*33.	In-place value of existing pavement/compacted C.Y.
*34.	Existing pavement salvage value as percent of present value
*35.	Level-up required for the first overlay
*36.	Number years existing pavement has been open to traffic
(Fie	lds 37-44 for each material considered in design)
37.	Layer designation number

TABLE 2. (Continued)

	Sample PhaseDatDesign As-built SubsequentIf Othe	a Source r Than PFDS
38.	. Letter code of material	
39.	Name of material	
40.	. In-place cost/compacted C.Y.	
41.	. Stiffness coefficient	
42.	. Minimum allowable thickness in initial construction	
43.	. Maximum allowable thickness in initial construction	
44.	. Material's salvage value as percent of original cost	

.

τ.

It may also be pertinent to note that other climatological data may be obtained just as easily. Rainfall, wind, and evaporation are three such factors that could eventually become inputs to a new environment effects submodel within the pavement design system.

The other three factors in the environmental file all contribute to the swelling clay model that estimates performance degradation due to subgrade movements. These factors are unique to FPS-11 and must be acquired and contained in PFDS.

- (3) Performance File: contains those factors essential to describing the pavement performance. The first two factors are self-explanatory; they are fundamental to any measure of performance. Minimum serviceability index and design confidence level are shown to fall in subsequent data phases, or sometime during the life of the pavement. This possibility could arise only if these design parameters are changed in a subsequent overlay design.
- (4) Traffic File: contains the essential traffic data for design. Note that all of these are shown to exist in the THD D-10 traffic file; therefore, no data storage in PFDS is necessary. This file will consist only of a computer instruction to access the D-10 Traffic File by its proper name.
- (5) Costs File: these three factors are the only ones for which as-built or subsequent data collection appears appropriate.
- (6) Constraints File: this file will be a simple listing of all of the remaining FPS-11 inputs not already a part of another file. These data factors are the program constraints and judgment values of the designer. They have a design value only and will be used to reconstruct the design strategy at a later date.

Management Factors

The management potential of the proposed PFDS is somewhat difficult to predict, but the one most important management factor is clearly serviceability index. Different factor combinations will undoubtedly be used by each District Engineer to satisfy his own particular needs. Service life estimates derived from design and as-built data should serve adequately for construction programming and budgeting purposes.

One other management factor not included in PFDS merits mention here. The Skid Resistance Factor is a measure of increasing importance in the interest of highway safety. It indicates when some type of maintenance action or surface treatment is needed to improve skid resistance, and D-8R has begun an extensive program of testing and recording this information. Like D-10 files, this file will be accessible in MARK IV by its proper name as needed; there is no apparent need to store skid data in PFDS files.

Special Research Files

Most of the special research projects discussed earlier will involve acquisition and analysis of varying quantities of data. It may be advantageous to utilize computer processing of such data and consideration should be given to definition of temporary files within PFDS to serve the research project. MARK IV makes creation of temporary and working files relatively simple and this capability should be exploited fully.

CHAPTER 6. EXISTING DATA FILES AND SOURCES

Scope of Existing Files

With the essential data factors for PFDS identified, it is now logical to examine the existing data assets of THD to ascertain how many factors are already included in current files. Over the years, the Texas Highway Department has acquired a wealth of highway data and developed a comprehensive set of files for many purposes. When the computer came along, the Planning Survey Division (D-10), along with the Automation Division (D-19), promptly set about automating the data handling procedures so that today, practically all of the D-10 files have some form of computer handling. J. E. Wright, THD D-10 Division Head, announced the Texas plan for a state-wide data base at the Highway Research Board symposium on automation in August 1971 (Ref 91). Over three years before, D-10 had completed a thorough system analysis and published a two volume report (Ref 78). By December 1971, D-10 was operating 10 data files consisting of over a half million records and another 350,000 accident records were being processed each year.

D-10 Data File Relevancy

An early PFDS research objective was to thoroughly examine these D-10 files and identify data needed for the pavement design and research process. This proved very profitable as was indicated in the preceding chapter. Many of the essential factors are already in one or another of the D-10 files and the problem then becomes only a matter of how to access such files and fields. This problem appears well on the way to solution with the acquisition of the MARK IV file handling system and the potential for a universal record key discussed in Chapter 4. The remaining obstacles are discussed in Chapter 8.

Although identification of the specific essential factors that could be accessed from D-10 files was made in the preceding chapter, it is doubtful that the scope is adequately defined without a look at the files and their contents. Two D-10 files are highly relevant to PFDS:

- (1) Traffic Log with approximately 66,500 records and
- (2) Road Life (RL-1) with approximately 161,000 records.

The Traffic Log. Other than the basic highway identification information, there are 31 different fields of data in this file. Ten fields are Annual Average Daily Traffic (AADT) for the preceding 10 years. This information is derived from both automatic and manual counting stations throughout the state. The remaining data items are essentially for future design functions and are derived from loadometer studies as well as computation. Included in this group are traffic growth factor, 18 kip single axles, and percent trucks. All of the data items used in a standard THD pavement design procedure (Ref 74) are contained in this file, and are made available to other divisions and districts upon request. Data records are coded with the beginning and ending milepoints, so the specific segment of a given control section can be readily pinpointed. In summary, this file is already used as an important information source for the pavement engineer and will continue to be important (see coding form and sample output, Appendix C).

The Road Life File (RL-1). This file is probably the most relevant, comprehensive, and consolidated source of information that the THD pavement engineer can find anywhere in the Department. Furthermore, examination of the manual records from which the computer file is derived will show that they have been diligently maintained. This would suggest a high degree of data accuracy.

The system analysis staff report (Ref 78) states that the RL-1 records are "in detail as to type and design of construction and the dollar investment by construction components." The following discussion will attempt to summarize how comprehensive that statement is as applied to the pavement design function.

The manual records consist of the penciled form RI-2, "Log Record of Project Construction and Retirements." A typical record was selected at random and is shown in Fig 11. Note that the control section information is prominently displayed in the upper right corner; each control section is a separate record. The local features describing the termini of the control section are given on the top left. The next significant feature is the job number, the fourth column from the left. This record depicts every item of work performed on this segment of roadway since 1927. Each of the parallel



Fig 11. Typical road life record.

bars upon which shading and notations have been made represents the same roadway but shows only that work done on the designated job number. There have been 12 job numbers but only 10 are accounted for here. The other two apparently had no effect upon the pavement. A number of codes are used but note the numerical data on depth, width, and type of surface, base, and shoulders.

The automated version of this data record is equally impressive. Using rather extensive coding, the records can contain up to 132 different data items per segment of roadway as defined by control section and job number. The following is a selected list from the RL-1 file that are most relevant to PFDS (Ref 77) (see Road Life coding form, Appendix D):

- Surface type 13 different codes from bladed earth to ACP overlaid concrete.
- (2) Type of work 44 different codes used in combinations to allow designations from simple grading to complete reconstruction.
- (3) Cross Section of Surface width is coded to nearest foot; thicknesses are coded to nearest tenth of one inch.
- (4) Cross Section of Base coding tolerances same as for surface.
- (5) Shoulder Type 10 different codes from no shoulders to curb and gutters.
- (6) Shoulder Width coded to nearest foot.
- (7) Type of Treatment (Surface) 15 codes to designate the range from none to hot plant mix asphaltic concrete.
- (8) Aggregate Distribution coded to specify number of square yards covered by one cubic yard of surface aggregate.
- (9) Base Material 9 different codes none to brick.
- (10) Method of Base Stabilization 8 codes to designate range from none to lime, asphalt, and cement.
- (11) Subgrade Material 3 codes: earth, flexible base or foundation course, and select material.
- (12) Subgrade Stabilization Method 4 codes: none, cement, asphalt or lime.
- (13) Subgrade Width coded to nearest foot.
- (14) Subgrade Depth actually means subbase and is coded to nearest inch.

From the above, it is obvious that the cross section of the existing pavement structure (plan quantities) can be easily developed. It must also be noted that while concrete pavement surface was omitted from the above discussion, there is comparable information in the file for that material as well (see data fields 21-24, Appendix D).

In addition to the above pavement structure data, the file contains interesting information on the <u>reason</u> and <u>method</u> of pavement retirement, along with the length, date, and cost of the retired pavement. The term "retirement" as used herein means removal of the pavement from service for one or more of several reasons. For example, the pavement may have failed or become obsolete. RL-1 allows 28 different coded <u>reasons</u> for retirement. <u>Methods</u> of retirement have 51 different coding options varying from simple resurfacing to reconstruction along new line and new grades. This type of information is analogous to the design comments specified as an essential factor in the Structural File of PFDS.

The net potential of the RL-1 file is to provide data for all but two of the essential factors in the PFDS Structural File as indicated in Table 2. This data could initially suffice for the design sample phase until a reconstruction design is prepared using FPS-11 or its successor program. However, one significant problem exists in regard to this data file. Specifically, data records are set up for each job number, but the physical limits of the job number are not contained in the file. This means that there is no direct measure of the beginning and ending points referenced to some common base such as milepoint or beginning of control section. As previously pointed out, the traffic log contains these essential data.

To be of full value to PFDS, this type of locational parameter must be incorporated in the file. D-10 has given this matter consideration in the past and decided that insertion of the milepoint data for all previously accomplished projects was simply too large a task to be practical. It should be emphasized that milepoints for each job number could be derived from the manual RL-1 records; each job number is defined on the final plans with stationing referenced to some known point such as beginning or ending of control section, geographical feature, or political boundary.

It is suggested that a limited effort could provide a reasonable solution to this problem. D-10 has already reserved columns in the RL-1 file record format for beginning and ending milepoints. Therefore, no format changes are necessary and a coordinated effort between D-8 and D-10 could be undertaken

- (1) begin coding beginning and ending milepoints for each new job number being entered in RL-1, and
- (2) select on a case by case basis certain previously accomplished major construction/reconstruction or overlay job numbers to compute the milepoints (or stationing) from penciled RL-1 records and enter in the automated file.

For the immediate future, the above two steps can probably be done manually as needed with acceptable efficiency. When MARK IV is fully operational with a random access capability, the data fields in RI-2 can be accessed to yield certain milepoint data upon which calculations for RL-1 data can be automatically performed. For example, where control section milepoints are already entered in the road inventory (RI-2) files, a processing program could be written to take the job number stationing from RL-1 input, access milepoint from RI-2, compute job number milepoint limits and store the result in the appropriate record files in RL-1. In any event, the existing deficiency of detailed locational data for job number in RL-1 can conceivably be eliminated on a phased basis with little total increase in existing workloads. The amount of extra work is acknowledged to be an important consideration and should be carefully weighed in any adopted course of action.

Another limitation of the RL-1 file is the existing provision for only two subsurface layers. Data describing the surface and base layers appear completely adequate and the data entered in the subgrade field (field 28, Appendix D) are descriptive of the subbase. When more than one subbase layer is used in construction, all are combined as one composite layer for recording purposes. In some instances, two markedly different materials could be used for subbase layers and the data entered may not be sufficiently descriptive for both layers. On the other hand, a continuously reinforced concrete pavement (CRCP) overlay of an existing jointed portland cement concrete (PCC) pavement with a bond breaker may be considered as different layers. D-8 and D-10 should evaluate this restriction and determine whether there is sufficient merit in establishing more data fields to accomodate additional subbase layers.

Other Data Files

The automated data file Road Inventory (RI-2) previously referred to could be relevant to pavement design functions and is certainly an excellent management tool. RI-2 contains broad summaries of the structural composition besides highway grades, curves, and administrative information. In some respects, it is representative of the other more detailed files of traffic, structure, and geometrics. It has the disadvantage of being a sequential tape file as is RL-1 and random access of the data records is impossible (see Appendix E).

For environmental information, the National Weather Service (NWS) possesses records that are readily accessed by computer to yield a wide variety of data factors. The ones essential to PFDS are maximum and minimum daily temperatures, from which the district temperature constant is derived. As discussed in Chapter 5, this data is readily available for the past 22 years for all 1,000 NWS temperature stations in Texas. In addition, the following data are available from the same source for the period 1948-1970:

- (1) daily and monthly precipitation for 1,400 stations,
- (2) daily and monthly pan evaporation for 600 stations, and
- (3) daily and monthly wind movement for 300 stations.

The acquisition of only the temperature data tapes is suggested for the present. These should be used in a special research project to evaluate whether temperature constants vary sufficiently within the district to warrant use of "localized" values. If not, there seems little point in transferring some 16 million temperature readings to a computerized direct access storage device, even though it would be a simple matter to do so. Frank Scrivner, Texas Transportation Institute, (Ref 66) computed the district temperature constants now being used in FPS-11 from 10 years of data, and appreciable departure from such values seems highly unlikely, even with over twice the data. Furthermore, Ramesh Kher et al found in Project 123 sensitivity studies that the factor is only a moderate influence upon optimal design strategy (Ref 47). Such a special research effort could also conclude whether future acquisition and analysis of temperature data are warranted and if so, in what summary form they should be stored.

In the area of manual data files, the scope is simply too large to even summarize. Besides the vast files of the Equipment and Procurement Division (D-4), the individual districts possess a wealth of information from the construction control process. These records may become important in the development of sampling processes since there may already exist adequate procedures

for <u>acquiring</u> the data. Sampling may then reduce to a problem of selective <u>reporting</u> via remote computer terminals according to some specified format. This is discussed further in Chapters 7 and 8.

Summary

The D-10 Traffic Log and Road Life (RL-1) file are by far the most important records of immediate interest. Hopefully, THD acquisition of MARK IV coupled with redefinition of these files with compatible record control keys will make the data immediately available to the pavement designer and manager. It is emphasized that these records appear to be reliable and complete and it is suggested that they be used to the maximum extent. Furthermore, where changes and/or additions are desired to some data fields, a first recourse should be to work through existing procedures to obtain satisfaction. Just as with the record control key, the emphasis should be on making existing mechanisms work unless clear-cut deficiencies dictate otherwise. The great value of such efforts can be seen from the milepoint deficiency in the RL-1 file; the vast and relevant data elements therein can be had for relatively small additional effort.

CHAPTER 7. DATA SAMPLING

Sampling Concepts

After the fundamental decision is made as to what data to collect, the next step is recognition of the sources of data. Except for the assigned and computed data values, all other data must be obtained from either accessing other data files or sources, or measuring the factors in the field. Either way, the pavement engineer must be confident that the numbers he gets are valid representations of the pavement design problem. This means that a data sampling plan or philosophy must be developed and applied to both parts of the problem. For data to be accessed, the sampling plan and procedures must be checked; for data to be collected, the plan and procedures must be developed.

Sampling Techniques

Consultation with statisticians reveals that the best practical sampling approach for this type of data system is a systematic random sample on a stratified basis. Loosely translated, this simply means that the highway from which data is to be measured is first separated into strata or levels that are fairly homogeneous within themselves. Then each strata is sampled by beginning at a random point and taking sample values at regular intervals throughout the strata. As an example, serviceability index (SI) may be our particular pavement factor. First we select a segment of highway that has the same type surface and was probably constructed by the same contractor at the same time (same contract); this is our stratum. Then we pick a random starting point and measure and record an SI value every 2,000 feet throughout our selected segment.

Two significant features of the stratified, systematic random sampling method should be explicitly recognized (Ref 10, pp 206-230):

(1) There should be no cyclic effects within a stratum. For example, seasonal changes are a cyclic effect upon climate and any factor dependent upon climatic conditions must be sampled so that appropriate representation of each season is obtained. In some areas

of Texas, the surface curvature index (SCI) is directly dependent on time of year. SCI readings should then be taken within each season.

(2) The variance tends to be overestimated, i.e., larger than the true variance. In general, this means that probabilistic design values derived from such data should be conservative, i.e., there should be a kind of safety factor.

The same sampling procedure is applicable to time variable data as well. After selecting the strata of the factor, each stratum is sampled at a regular interval of time, i.e., each year. The same example of SI may serve here; SI values will be taken each year at some particular time.

This time interval of sampling will be, of course, dependent upon capability to take the sample. Presently available THD resources may not permit annual sampling. For example, THD has four Mays Road Meters for taking SI readings. Simple arithmetic shows that to cover the 70,000 miles of state system in one year, each unit must cover about 70 miles per day. This kind of schedule may not be possible for several reasons and the plans would have to be changed to fit equipment, or vice versa. However, implementation of PFDS will be a phased procedure and will not require immediate sampling on all highways. Therefore, sampling capability can be matched to needs on a progressive basis.

Stratum Equals Job Number

In Chapter 4, a record control key was defined to include the job number of the particular work being performed. The definition of job number shows that it is variable in length and scope of work but is generally a consistent type of pavement work, i.e., maintenance, overlay, reconstruction, or construction. In general, it will all be done by the same work force (THD work force or contractor) at one given period of time. If it involves reconstruction or construction of a new pavement, the same design thickness and materials will probably be used throughout the length of the job. There are a number of other variations such as number and type of superstructures, but these generally have no bearing on pavement construction. The net effect is that job number as a unit of work measure becomes a natural and practical homogeneous stratum for sampling purposes as far as pavement structure is concerned. This was a consideration in the selection of job number as an element in the record control key and the result is a maximum benefit from a consistent method of highway identification in the existing Texas Highway Department records system.

There will undoubtedly be instances where distinct breaks in pavement characteristics or design will occur within the length specified by a single job number. For this reason, a decimal position has been allotted to the job number to permit up to nine subdivisions of a given project (job number). This is emphasized to apply to such situations as a change in design thickness of pavement, not to stochastic variability expected within a project.

Sampling as a Function of Objective

Sampling must be guided by the purpose for which the data will ultimately be used. PFDS is expected to contain data for

- (1) management,
- (2) design, and
- (3) research.

For management, a single representative data value may adequately represent a factor in a control section and job number segment of roadway. Design will probably require more data values to permit selection of a critical value upon which to base a design. The term "critical" is used here in the statistical sense and refers to that value which gives the engineer a specific level of confidence that his design will be adequate. Research undoubtedly needs more detailed information than either of the other two objectives. In the beginning, the essential factors are probably a common denominator for all three objectives and were selected accordingly for PFDS. This will change with time; as design models prove adequate for a situation, they may be moved from the research to the design realm. Management factors will probably always be highly summarized values of a few basic factors.

Sampling as a Function of Resources

Each essential factor must be looked at individually to establish its desired sampling density. Now another consideration comes into focus, namely, the availability of THD personnel and equipment resources. This was briefly introduced in an earlier discussion of serviceability index sampling. Equipment
available will be a fundamental limitation in many cases. In other situations, the number of qualified personnel available will govern. This will be influenced greatly by the policies and interests of the particular District Engineer. In fact, the District Engineer may determine the success or failure of the data system in his own district. Therefore, it is especially important that he be given all the facts and potential in the most concise, accurate, and direct manner.

Standardized Measuring Apparatus and Techniques

In order that data from district to district may be directly comparable, it is important that the measuring apparatus and techniques be standardized. For example, it would be undesirable for one district to use a Mays Road Meter to sample serviceability index while another uses a rating panel. Similarly, the techniques for arriving at triaxial class should be the same for all districts. In some cases, this is going to require a great deal of coordination. However, much has already been done. A method for calibrating the various Mays Road Meters has been developed and is being refined now (Ref 85). For many years, the quality control tests for materials have been standardized and codified. This is well illustrated by the job control and progress record testing required in the Texas Highway Department by the construction manual (Ref 72). Figures 12 and 13 are taken from that manual and show the guide test schedule for asphaltic concrete pavements, surface treatments, and the base, subbase, and embankment materials. It will also be noted that the concept of systematic sampling discussed earlier in this chapter has been followed in the illustrated sampling guides. In summary, it appears that most of the established sampling methods will probably suffice for PFDS and the problem is reduced to standardizing tests for those essential factors that are new to the design engineer and were first introduced to the pavement design field by the flexible pavement system (FPS). Such factors number only six:

- (1) serviceability index,
- (2) stiffness coefficient,
- (3) subgrade swelling probability,
- (4) subgrade potential vertical rise,
- (5) subgrade swelling rate constant, and
- (6) surface curvature index.

EMBANKMENTS, SUBBASES, AND BASE COURSES

	ELE TONIC CODING AND COLLEGIST		JOB CON	TROL TESTS	PROGRESS	RECORD TESTS	FINAL	
	TEST DOR	ILST NUEIPER	LOCATION OF TIME OF SAMPLING	TREQUENCY of SAMPLING	LOCATION OF TIME OF SAMPLING	FREQUENCY of SAMPLING	RECORD TESTS	REMARKS
	Ger pitchan	Tex- 114-L	As Designated by	Each 3,060 C.Y.	Sume as Jon Control	Each 100,000 C.Y. Min, J	Progress Record Tests Are Also Final Record Tests	
	Gradation	Tex- 110-E	During Stockpiling Oprs. from Stockpile, or from Windtow	' Each • F,200 G,Y, or L,000 Tees	Some ds Jop Control	t Out of 20 Job Control Tests Min. 3	Per Trivet-Way Per Mile Min, J-Max, 10 (6)	Final Recent Tests For Informational Purposes Only
	Liqvid Limit	Tex- 304-E	During Stockpaloig Oprs., from Stockpile, or from Winnow	Cach 1,200 C.Y. or 1,600 Test	Same as Job Control	t Out of 20 Jon Control Tests Min, 3	The Havel-Way Per Mile Min, 3-Max, 10 (P)	Final Record Tests For Informational Purposes Only
Unses.	Flasticity Index	Тгж- 106-Е	During Stockpiling Opris, from Stockpile, eastern Window	Each 1,200 C.Y. or 1,600 Fon-	Same as Job Control) Out of 26 Job Control Texts Min, 3) Per Davel-Way Per Mile Min. 1-Max. 10 (P)	Finil Record Tests For Informational Purposes Unly
ATED S MSC - 20	Wet Ball Mill	Trx- 116-E	During Stockpilling Opris. from Stockpile, or from Winnew	Each Normal Dey of Production				
UNTRI. ALD B	Triexial	Tex- 117-1.(11)	During Stackpilling Opts, from Stockpile, or from Windraw	Lach 12,000 C.Y. at 16,000 Tons				Not a Field Laboratory Function
	Compaction	Tex- 114-E	As Designated by The Engineer	Each 2,000 Lin. Ft. Per Course Per Travel-Way (D)	Same as Job Control	1 Out of 20 Job Control Tests Min. 3 (C)	Progress Record Tests Are Also Final Record Tests	
	Thickness		As Designated by The Engineer	1 Depth For 2,000 Lin. Pt. Per Course Per Travel-Way			Total Depth Fer Travel-Way Per Mile Min, 3-Max, 10 (R)	
BACL MATURIAL	As Shown Above Fer Untreated Base							
UNE STREES	Compliance With Item 264 "Hydr. Line & Line Slurry"	Tex- 600-1	Sampled, Tested, and Approved by D-9		-			
VD IWSE TO CEMENT	Compliance With The Std. Specifications & Spl. Provisions	лятм С-150	Reilroad Car, Truck, or Cement Sins	Each 2,000 Bbla. (May Be A Compesite Sample) For Each Type or Brand	Same as Job Control	Witness Sampling At Least Once		Each Brand and Each Type To Be Sampled and Tested Separately
BBASE AN	Compliance With Item 300 "Asphalts, Oils and Emulsions"	Tex- 500-c:, Etc.	Sempled, Tested, and Approved by D-9					
	Gredation (A)	Jex- 110-E 6 Spec's.	Roadway; After Polyerization	Each 1,200 C.Y. or 1,640 Ton.	Sanie as Job Control	i Out of 20 jnb Control Texts Min, J (C)		(A) Where Required To Control Descention Followertexticat
TRC ONFEET	Compaction	Tex- 114-E	As Designated by The Engineer	Each 2,000 Lin, Ft. Per Course Pet Tracol-Way (8)	Same as Job Control	1 Out of 20 Jub Control Tests Man, 3 (C)	Progress Record Tests Act Also Enal Record Tests	
0	Thickness		As Designated by The Engineer	1 Depth Per 2,000 Lin. Ft. Per Course Per Travel-Way (B)) Total Depth Per Travel-Way Per Mile Min, 1-Max 10 (b)	

Distinct Engineer Will Select Any One of These Three (3) Locations of Any Combinations Thereof With The Provision That A Minimum of 25% of Tests Will Be Sampled from The Windrow For Gradation, Plasticity Index, and Liquid Limit Only.

.

SURFACE TREATMENTS

AGGREGATE	Gradation	Tex- 200-F (Dry)	At Source or At Point of Delivery	Each 300 C.Y. Min.)	Same as Job Control	1 Out of 20 Job Control Tests Min. 1	Record Testing Not Required Where D-9 Provides Job Control Testing
ASPHALT	Compliance With Item 300 "Aspnalts, Oils and Emulsions"	7ex- 500-C Etc.	Sampled, Tested, and Approved by D-9				

Rev., Oct., 1966

Fig 12. Materials sampling guide schedule (after Ref 72).

ASPHALTIC CONCRETE PAVEMENTS

ANNINUM SAMPLING AND TESTING PER CONTRACT		JOB CONT	ROL TESTS	PROGRESS	RECORD TESTS	FINAL		
IA D.E IAL PRODUCT	TEST FOR	TIST NUMBER	EOCATION of TIME of SAMPLING	FREQUENCY RESECTING	LOCATION OF TIME of SAMPLING	FREE PENCY of SAMPLING	RECORD TESTS	REMARKS
	Gredation	Tex- 200-F (Dry)	During Delivery to Plant or From Stockpile	Each 6,000 Tons	Same as Job Control	Each 60,000 Tons Min, 2 (A)		Testa To Be Made for Each Stockpile
COAKSE	Deleterious Material and Decontation	Tex- 217-F	As Designated by District Engineer or as Specified	Each 6,000 Tons	Same as Job Control	Each 60,000 Tons Nin, 2 (A)		
V	Plasticity Index	Tex- 105-E	As Designated by District Engineer or as Specified	Each 6,000 Tons When Necessary	Some as Job Control	Each 60,000 Tons Min. 2 When Neccessary (A)		
at cante	Gradation	Tex- 200-F (Dry)	During Delivery to Plant or From Stockpile	Each 6,000 Tons	Same as Jub Control	Each 60,000 Tona Min, 2 (A)		Tests To Be Made for Each Stockpile
A 3CRU	Plasticity Judex	Ti x- 106-E	During Delivery to Plant or From Stockfile	Each 6,000 Tons	Same as Job Control	Lach 60,000 Tuns Nin, 2 (A)		
VINGALL	Gradation	Теж- 200-1 (Тлу)	During Delivery to Plant of From Stockpile	Each 6,900 Tong	Same es Job Control	Each 60,000 Tons Min. 2 (A)	N	T
aŭ	Gradation	Тох+ 200-Г (Diy)	Hot Bias	4 For Each Normal Day's Production	Some as Jub Control	Each 30,000 1		Mineral Filler Not To Se Included
ALECON	Sand Equivalent	Tex- 203-F	Hot Bins	1 For Each 10 Days' Normal Production	A			
0 ¥				- And -				
NONS	Compliance With Rem 100	Inv		V				
STO NILLS	"Asphaits, Oils and In-	10x- 207-F	Plant or Read	1 For Dach Normal Day's Production	Sane as Job Control	1 For Each 10 Days Normal Production (A) Mine 2		
-	Stability	Tex- 208-5	Plant or Ro-H	L For Lach Normal Day's Production	Sume as Job Control	I For Lach 10 Days' Normal Production (A)		
P.6	Extraction	T++x- 210-F	Plant or Read	t For Each Nurmal Dey's Production	Same as Job Control	1 For Each 10 Days' Normal Production (A) Min. 2		
T MB AC	In Place Density (When knowned by Specifications)	Тех- 2117-г	Completed Course) for Fach Normal Day's Production	Same as Jrw Control	i For Each 10 Days' Normal Production Min, 2		
DT MBC	Culture tometon (Willow Required by Specifications)	Tex- 214-F	Piant or Road	L For Each Normal Day's Deduction	Same as Job Control	Tor Each 10 Days' Normal Production (A) Min. 2		···
Ĭ	Musstare Content	Тея- 712-Е	Plant or Road	i for Each Normal Day's Production (C)	Same as Job Control	i for Lich 10 Days' Normal Production (A) Min. 2		····
	Hydrocurbon Volatile spotent	T0X- 133-t	Fiant or Road	1 For Each Normal Day's Projuction (C)	Semt et job Centrul	Normal Freduction (A) Min. 1		
CULD MD	Compliance With Num 330 "Cold-Mix Linustone Rock Asphals"	Tox- 217-F Etc.	Semplust, Testod, and Approved by D-9					
	Dimensions		Completed Pavement	As Necessary			I Depth & I Width Per Travol-Way Per Mile (b)	<u>,</u>

Rev., Oct., 1956

Fig 13. Asphalt surface sampling guide schedule (after Ref 72).

Factor (1) has received much attention and procedures for data acquisition are well developed (Refs 31, 84, and 85); procedures for factors (2) and (6) are standardized to a degree in Ref 73; Lytton discusses methods for quantifying factors (3), (4), and (5) in Ref 53. A forthcoming revision to Ref 73 is expected to clarify the procedures for acquiring swelling clay parameters and should be consulted in conjunction with the theory presented in Ref 53.

Future Action

Because the THD districts are highly autonomous, it is imperative that one or more of them be selected for PFDS pilot implementation on the basis of their interest. The plan of action should then be:

- (1) Review the array of essential factors proposed herein and add those meeting some agreed essentiality criteria. On this point, it will be vitally important for someone to constantly stress the importance of eliminating data as discussed previously or data volume will grow without bounds.
- (2) Examine existing district data collection plans to determine how many PFDS essential factors are already being sampled.
- (3) Evaluate the existing data collection procedures to be satisfied that sampling is consistent with concepts previously discussed.
- (4) Device sampling plans and procedures for the remaining essential factors.

With the new remote computer terminals in the district headquarters, the option of inputting data via punched cards (or some other medium) should be explored with D-19. However, this action should probably be deferred until MARK IV is fully operational and D-19 has had time to publish some basic operating procedures.

This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

CHAPTER 8. FILE STRUCTURE AND DATA ANALYSIS

Overview of System and File Structure

This chapter was written for the pavement engineer, not the computer programmer or systems analyst. The latter two may even cringe at the liberties taken herein with computer systems terminology, but the object is to clearly depict to the practicing engineer some basic features of the proposed PFDS files. It is readily acknowledged that some of the resulting illustrations do not reflect the precise form of the computer handling of the data but it is contended that there is no harm done by looking at them in this way (see later discussion titled "File Combinations").

Another important point to emphasize is that this discussion is generalized and applies to no particular data handling system or language. The MARK IV file handling system now being introduced in the Texas Highway Department has its own way of defining, creating, and processing files, trailer files, temporary files, etc. The options in MARK IV far exceed anything envisioned herein, so the computer programmer will be able to easily define the desired file manipulations with the system.

The basic concept of the PFDS file structure is to store all design values in "prime" files and all as-built and subsequent detailed values in "trailer" files. The trailer files can be processed individually or together with their prime file (Ref 42). Each of the PFDS files as described in Chapter 5 will be illustrated and described in order to make the operating concept clear.

First, however, an overview of the file structure will be helpful to identify major groupings of data fields. Figure 14 shows both the conceptual prime and trailer file structure. Note that two differences are illustrated:

- (1) The record control key for the trailer file is longer than for the primary. Since the records in the trailer file are detailed components of the prime file records, it is necessary to extend the data definition.
- (2) The data fields are fewer in the trailer file. This simply illustrates that, in general, not all factors present in design need additional sampling in time and space. Therefore, only those factors to be further sampled are found in the trailer file.





Fig 14. Overview of conceptual file structure.

There are some minor deviations from the structures shown in Fig 14 and they will become evident when the files are examined. Essentially, the deviations amount to certain omissions of data fields not needed. This is particularly evident in the environment and costs trailer files.

One objective in structuring these files has been to preclude as many "embedded" empty data fields as possible. This is accomplished primarily by utilizing variable length records that are extended or collapsed to fit the requirements of the particular record. For example, if a data record contains data on a two-layer pavement, no provision is made for a nonexistent third layer that might ordinarily exist. This should conserve storage space although storage for PFDS will probably not be a problem for a long time.

Data Coding

A determined effort has been made to keep data coding to an absolute minimum in order to keep the system input and output as readable as possible. However, a certain number of codes appears desirable and they have been devised to be as logical as possible. They apply to all files shown in Figs 15 to 22. Coding used in only one file is defined in the legend on the corresponding figure. The following definitions and codes are applicable:

- Sample phase: designates the time when data values are to be acquired as indicated also in Table 2. The following numbers designate
 - 1. design values
 - 2. as-built values
 - 3., 4., 5., etc. subsequent measurements in time.
- (2) Date: always a 4-digit number, the first two digits being year and the second two designating the month.
- (3) Layer designation and code: these entries are generally the same as those used in FPS-11; layers are designated 1, 2, 3, etc. and materials A, B, C, etc. as part of the program input. Two additional codes that may be used in this field are: OL = overlay and SC = seal coat .
- (4) Sample number: sequentially assigned numbers within a sample phase to identify number of the sample value. These numbers uniquely identify the sample with date and milepoint and may go to 99.

Decimal points have been omitted in most of the illustrations to conserve

space. This is analogous to the assumed decimal points in the actual data files to conserve storage space.

Structural File

Figures 15 and 16 depict the prime and trailer structural files. Representative numbers have been entered to illustrate how the file records would be collated; dashes indicate that the entry in the preceding record is still valid in that field and to help accent record changes.

Note that in Fig 15 each job number is one record in the prime file, and all the pertinent design data appear in that one record. The result is a very long record; for a three-layer design with two subsequent overlays and design comments limited to 80 characters, the record length would be over 200 columns. This length could be reduced by shortening the names of materials to an abbreviation, etc. However, there may be many instances where the design comments should be longer than 80 columns. It is therefore suggested that all of the comments be designed into trailer records as described by Ref 42 and equated in the MARK IV system. This should pose no difficulty for the programmer.

In the trailer file depicted in Fig 16 there can be several records for each job number because of the need for specific date and milepoint sampling. For this reason, the record control key must have two additional fields to uniquely identify the record. As pointed out in Chapter 4, the record control key must be given exactly to the system or the record cannot be located in a random access environment. Since milepoint computed to three decimals may be difficult to recover at a later date, it is not a good key component. Therefore it is proposed as an item of data since its presence in the file is essential.

The multiple samples for any job number define the variability inherent in construction practices, the principal objective of this file structuring. Note that this trailer file should always contain the as-built data, sample phase code 2.

Description of one series of the illustrated data examples should help to clarify these and subsequent figures. In Fig 15, the first record shown is job number 1550 in control 2374 section 15. The design of this project was an asphaltic concrete surface five inches thick and a gravel base 12 inches



Notes: (1) Each row equals one record.

(2) In programming, a count field will be inserted before each repeated series of fields, i.e., before subsurface layers and events.

Fig 15. Structural prime file.



Note: Insert count field before first subsurface layer.

Fig 16. Structural trailer file.

thick. Only one overlay is illustrated at 5.5 years after initial construction. The next record is for job number 570, and contains the design values for the overlay predicted in job number 1550; note that it was designed exactly six years after the initial pavement construction.

The first three records illustrated in Fig 16 are individual as-built samples from the initial construction of job number 1550. The next three records show as-built samples taken from the overlay constructed five months after design and five years and five months after construction of the initial pavement. This latter value corresponds to the 5.5 years predicted in inital design.

The horizontal rows of dots between records in all figures represent records omitted. In Fig 16, an unspecified number of additional as-built samples from both the initial construction and overlay have been omitted. Job number 1560 has also been omitted but will be seen to be a special maintenance project in the cost files (Figs 21 and 22).

Environment File

Figure 17 shows the proposed environment prime file structure with sample entries to illustrate how the records would be sorted. Again, this file contains only design data, sample phase code 1, and each record is a different job number. The trailer file is shown in Fig 18; note that the milepoint field has been deleted. Since swelling clay data are generally representative of entire projects, single observed values for swelling probability and swell rate constant for the entire project (job number) are probably all that is required. Milepoints for job numbers are contained in the prime file. Note also that the sample number field is not needed when the specific milepoint is not used.

A value for potential vertical rise is shown for subsequent sampling phases but not for the as-built phase, and this warrants a brief explanation. This is intended to depict a possible way of collecting meaningful data on this parameter. Specifically, potential vertical rise is a long-term effect and no new evidence on its value is likely to be uncovered during construction. However, during the life of the pavement, the vertical rise should be measurable and these values can be used to verify estimates made during design. Therefore, the subsequent (sample phase 3 and 4) values of vertical rise shown



.



Fig 18. Environmental trailer file.

in Fig 18 are cumulative values to be compared to value of 7.2 used in design (see Fig 17).

Since the temperature constant is assumed fixed within each district, there is no need to repeat values for it in the trailer file. However, if additional research shows localized temperature constants to be significant, there must be a field in the prime file for it and such has been provided.

Performance File

Figures 19 and 20 illustrate the prime and trailer performance files. These files conform very closely to the conceptual file plan illustrated earlier. The explanation of the design confidence level code is given in Fig 19. Since these data are point data for the most part, the trailer file must accomodate both time and space sampling as shown.

Costs File

This file deviates from the conceptual structure to a greater degree than the others. The prime file shown in Fig 21 has the typical record control key and the usual time and space identification fields. Each record is a separate job number containing design data (estimate of cost) as is true for other prime files. However, note that job number 1560 has data entries for routine maintenance only. This value is computed per square yard and will cumulatively compare with the design value of 0.13 shown as routine maintenance cost in the first record. Since all special maintenance projects, as well as overlays, seal coats, and construction, receive separate job numbers, the estimated costs for each will always appear as design values (sample phase 1) in this file. The trailer file in Fig 22 has the usual record control key minus sample number, since no milepoints are needed. To simplify these data items, a new field entitled type work has been introduced; codes are as shown. Numerical entries are as-built pavement costs per square yard. These values are directly comparable to the costs shown for the corresponding job numbers in the prime file.

Traffic File

The traffic file is not depicted herein since it is anticipated that all necessary traffic data can be obtained from the D-10 files. The only things



Legend

Design Confidence Level:

A	=	50 percent	D =	99 percent
В	=	80 percent	Е =	99.9 percent
С	=	95 percent	F =	99.99 percent

Fig 19. Performance prime file.



Fig 20. Performance trailer file.





.

Legend

Fig 22. Costs trailer file.

needed are the name of the file and its structure; both are readily available from D-10. However, the data file is currently a sequential file (on tape). When the file is redefined in MARK IV, random access to the data will be possible. This step does not appear too far away and complete random access to the traffic data should be possible by the time PFDS is functioning in MARK IV.

Constraints File

The constraints file has not been illustrated since it is a simple listing of the remaining design values (FPS program inputs) for a project (job number). Contained herein are all factors that are designated as constraints in Table 2. The data factors shown as constraints in Table 2 will be sequentially arrayed in records with the standard key described in Chapter 4. This file is static and will be used only to reconstruct the design input listing (see Fig 25).

A Reference File

One reference file could be an extremely valuable asset for intelligent and efficient manipulation of the other PFDS data files already described. Specifically, a file that ties all job numbers to their specific control and section would be very helpful. In addition, some basic information about the job number could be included. Possibilities are

- (1) date performed,
- (2) type work,
- (3) milepoint limits, and
- (4) total cost (all work).

Inclusion of all of the above information might be a first step to automating the records in the control section office of D-8. Such action should be carefully considered by the personnel responsible. The total cost field would give PFDS users a bit of management information in highly accessible fashion. Insertion of the milepoint limits could make it possible to eliminate this field from the other PFDS files; this should be carefully evaluated by systems analysts to see if it would result in an acceptable file processing environment in MARK IV. For the present, it is suggested that the file be kept simple and limited to job number, date, and type of work. This file is illustrated in Fig 23. It makes possible the immediate retrieval of the essential



Note: In programming, insert count field before first series of job number fields.

Fig 23. A reference file.

information needed for interaction with the other PFDS data files. Only job numbers back to and including the last reconstruction of the entire control section need be entered for an adequate reference file.

File Combinations

Inclusion of the milepoints in the reference file, as discussed above, should be carefully evaluated in light of the increased file processing made necessary by such data deletion in other files. Perhaps such an action would conserve storage space at an unreasonable expense of retrieval and/or update time. The systems analyst must weigh this in an operational environment and make a judgment.

However, some reduction in media storage can be realized by judicious combinations of certain files having the same basic characteristics. Two such combinations appear reasonable:

- the environment, performance, and cost prime files (Figs 17, 19, and 21 respectively), and
- (2) the structure and cost trailer files (Figs 16 and 22).

In each of the above cases, the data fields are comparatively short and could be stacked behind each other in each record to allow use of the same record control key field and time and space identification field. It is strongly suggested, however, that this computer programming action be accomplished without complicating the file use for the engineer. In other words, while the files may actually be combined as far as the programmer is concerned, the engineer should be able to interact with the system as though they were separate files.

Another programming aid might be mentioned here. As noted on Figs 15, 16, and 23, a count field is necessary in advance of repeated fields of data in records of variable length. This entry designates the number of times the data series is repeated in the file, i.e., the number of layers, job numbers, or events.

Prospects with MARK IV

When existing D-10 files are redefined in MARK IV, the files and fields will be given names that are needed to access the data. This will be true for PFDS as well. Ten standard forms are available to define and manipulate the data base in MARK IV and the instructions on use of them is a separate volume of its own (Ref 44). However, the action required by the average user should not be difficult to learn. D-10 and D-19 are currently studying this phase of the system. One especially powerful asset is the ability to change file structures with little effort. This is a very desirable feature in any system and especially attractive with a new data base.

Data analysis with MARK IV is somewhat limited although the operations possible are very easily invoked by terse entries on the standard form (see Fig 28). The following data manipulations are available:

- (1) Sorting of values in sequence, ascending, or descending.
- (2) Provision of summaries as follows:
 - (a) total value of items,
 - (b) cumulative value of items,
 - (c) count number of items,
 - (d) select maximum value,
 - (e) select minimum value, and
 - (f) compute average value.

Custom-Written Analysis Programs

There are some mathematical analyses necessary for PFDS that are not possible or practical with MARK IV. However, it is possible to access the MARK IV files with a custom-written analysis routine or module and perform the desired computations. The analysis module can be written in a high level language such as FORTRAN or PL/1.

Some rather elaborate analysis routines can easily be envisioned as desirable management tools. Event-triggered reports, for example, would be a great aid and might be typified by a serviceability loss prediction that is automatically triggered when the design life performance curve projects to a service life less than 85 percent of design life (or some other percentage). Such a computation could be automatically invoked each time a serviceability index, traffic, or surface curvature index value is input to PFDS. An illustration of this is given in Fig 24 where the as-built serviceability index was well below the design value of 4.2. The moment the as-built value of 3.9 is input to PFDS, the performance curve projection is computed by the analysis



Design Life - Years

Fig 24. An event-triggered report of serviceability life of pavement.

module and intersects the minimum serviceability index level just before the 17th year of service. The pavement engineer may then expect a higher cost of maintenance or plan an earlier overlay to prevent accelerated damage due to dynamic loads.

A good deal of imperfection in the existing models may make such analyses as described above somewhat premature. Perhaps, a better feeling for the distribution of data values is needed first. It is therefore suggested that the initial analysis routines be limited to the following:

- (1) A program to re-create the FPS-11 design inputs in the conventional order, as illustrated in Fig 25. This will involve extracting the design values from each of the PFDS prime files and combining them with the constraints file. This will permit a complete reevaluation of the design decision at a later date and allow direct comparison of design estimates versus actual performance.
- (2) A routine containing the following performance model (Ref 68):

$$N = \frac{Q\alpha}{KS^2}$$

where

N = 18-kip single axles Q = $\sqrt{5 - P} - \sqrt{5 - P_1}$

- P₁ = initial serviceability index
- P = present serviceability index
- α = district temperature constant
- S = surface curvature index
- K = regression coefficient
- (3) A routine with the swelling clay model (Ref 53):

$$P = P_1 - f(\alpha_1, \alpha_2, C_1, C_2, C_3, t)$$

where

P = present serviceability index
P₁ = initial serviceability index

TEXAS HIGHWAY DEPARTMENT FPS - 11 FLEXTBLE PAVEMENT DESIGN

PROB 18 ******	DIST. 14 **********	COUNTY TRAVIS	CUNT. 3136	SECT. 01 1	HIGH₩AY _P 1 ΜυΡΔC !********	DATE 12/28/71	LPE 238	PAGE 1
BASIC *****	DESIGN CRITE ********	R 1 A						
LENG MINI MINI MINI DESI INTE	TH OF THE AN MUM TIME TO MUM TIME BE MUM SERVICE GN CUNFIDENG REST RATE OF	NALYSIS PERIC FIRST OVERLA TWEEN OVERLAN ABILITY INDEX CE LEVEL R TIM ^C VALUE	DD (YEAR NY (YEAR YS (YEAR (P2 CF MONE	S) S) S) Y (PERCI	ENT)			20.0 4.0 6.0 3.0 E 7.0
PRGGKA *****	M CONTROLS /	AND CONSTRAIN ******	NTS ***					
NUMB Max Maxi Accu	ER OF SUMMAN FUNDS AVAIL MUM ALLOWED MULATED MAX	RY OLTPUT PAG ABLE PER SQ.N Thickness of Depth of All	DES DESI (D. FOR FINITIA DVERLA	RED (8 INITIAL L CONSTI YS (INC)	DESIGNSZP DESIGN (D RUCTION (I HES) (SXCL	AGE) OLLARS) NCHES) UDING LEVEI	L-UP)	9 03•6 36•0 36•0
TRAFFI *****	C DATA *****							
ACT ADT UNE- AVER AVER AVER PROP PERC	AT BEGINNIN AT END OF TO DIRECTION 2 AGE APPROACT AGE SPEED TO AGE SPEED TO CRTION OF AU ENT TRUCKS	G JF ANALYSIS WENTY YEARS I OYEAR ACCUN H SPEED TO TH HRUUGH OVERLA HROUGH OVERLA DT ARKIVING E IN ADT	S PERIOD VEHICLE MULATED HE OVERL AY ZONE AY ZONE EACH HOU	VEHIC S/DAY) NO. OF AY ZONE (GVERLA (NON-OV JR OF CO	LESZDAY) EQUIVALENT (MPH) Y DIRECTIC ERLAY DIRE NSTRUCTICN	18-KSA (N) (MPH) (CTION) (MPI (PERCENT)	34 64 6894(33C. 752. 5C.0 2C.C 5C.0 5.5 8.0
ENV IRU *****	NMENT AND S	UBGRADE ******						
DIST Swel Pote Swel Subg	RICT TEMPER LING PROBAB NTIAL VERTI LING RATE C RADE STIFFN	ATURE CONSTAN ILITY CAL RISE (ING CNSTANT ESS COEFFICIO	NT CHESI ENT					21.0 0.85 5.00 0.08 0.26
CCNSTR *****	UCTION AND	MAINTENANCE	DA TA ****					
SERV SERV MINI OVEA ASPH ASPH WIDT FIRS INCR	TICEABILITY TICEABILITY MUM OVERLAY LAY CONSTRU GALTIC CONCR GALTIC CONCR M OF EACH L T YEAR COST EMENTAL INC	INDEX OF THE INDEX P1 AFT THICKNESS (CTIUN TIME () ETE CUMPACTE ETE PRODUCTI ANE (FEET) OF ROUTINE N REASE IN MAIN	INITIAL ER AN OV INCHES) HGURS/DA D DENSII ON RATE MAINTENA NT. COSI	STRUCT VERLAY (Y) (Y) (TONS/H (TONS/H ANCE (DO F PER YE	URE /C.Y.) GUR) LLARS/LANI AR (DOLLAI	E-MILE), RS/LANE-MIL	EJ	4.0 3.9 0.6 7.0 1.20 75.0 12.0 100.00 10.00
DETOUR *****	DESIGN FOR	OVERLAYS *****						
TRAF TOTA NUME OIST DIST DETC	FIC MODEL U SL NUMBER OF SER OF OPEN SER OF OPEN ANCE TRAFFI ANCE TRAFFI DUR DISTANCE	SED DURING D LANES OF THI LANES IN RES LANES IN RE C IS SLGWED L IS SLDWED AKOLND THE	VERLAYIN E FACILI TRICTED STRICTED (CVERLAY OVERLAY	NG ZONE (O ZONE (DIRECT RLAY DI ZONE (N	VERLAY DII NON-OVERLI ION) (MILI RECTION) ILES)	RECTION) AY DIRECTIO ES) (MILES)	N J	3 6 1 3 1.00 0.0
PAV INC *****	G MATERIALS	INFORMATI ON ********						
LAY	MATER VER CODE	IALS NAME WT. ACP	COSI PER 0 21.42	T STR CY COEF 2 0.9	. МІМ. F. DEPTH 6 1.00	MAX. S DEPTH 1.00	ALVAGE, PCT - 10.00	

.

 $\alpha_1 = .335$ $\alpha_2 = .17$ $C_1 =$ swell probability $C_2 =$ potential vertical rise $C_3 =$ swell rate constant t = time in years

- (4) Routines accessing the stiffness coefficient and profile analysis computer programs explained in Ref 73. Care must be taken to insure that the data <u>names</u> in PFDS and these analysis programs agree. If PFDS is loaded on the MARK IV system, this will mean that the two analysis programs must be revised slightly to reconcile terminology.
- (5) A statistical program to compute mean, variance, standard deviation, and coefficient of variation.

CHAPTER 9. PFDS INFORMATION POTENTIAL

PFDS Equals Information

The fundamental purpose of PFDS is to provide the engineer manager with "information," not just "data." Charles Kriebel of Carnegie-Mellon University emphasizes the importance of distinguishing the difference (Ref 51). While data are simply various forms of facts, information is the intelligence of retrieved data when put to use in context. This important distinction is that the <u>information</u> system (PFDS) should require focusing on the <u>management</u> functions in addition to the pure data processing activities. It is the purpose of this chapter to illustrate how PFDS addresses the information needs of the highway engineer users. In some respects, it could be considered the most important part of this report.

It is appropriate to recognize three distinct groups of potential PFDS users:

- (1) the District Engineer and his staff,
- (2) the administrative headquarters and divisions, and
- (3) researchers.

Each of these groups will have a different type of information need. This was also discussed briefly in Chapter 7, Data Sampling, wherein the impact of data use upon sampling was considered. Throughout this report, an attempt has been made to emphasize the end use of data and the importance of building a data system responsive to the users.

Illustration of the potential use of PFDS by each of the above groups is given in the following sections. These illustrations are typical inquiries that might come from the engineer or manager and a typical response output from PFDS. It must be emphasized that the inquiry examples presented are not coded in any particular system except Figs 28 and 29; these two figures illustrate the MARK IV system. Emphasis has been placed on clarity rather than programming and coding accuracy.

Information for District Management

This user group is considered first for a very good reason. In most respects, the highway engineering job is being done by these field personnel and PFDS must therefore answer their needs first. As is pointed out in Chapter 10, the districts should be consulted early in the pilot implementation phase to insure that their needs will be satisfied.

The District Engineer is constantly faced with the judicious allocation of his men, money, and material resources to maintain the best possible highway system for the traveling public. PFDS can help him do this.

Serviceability index (SI) and skid resistance coefficient are two of the most important parameters to indicate how well a highway pavement serves the public. Therefore, when either or both of these factors begin exhibiting minimum values, it is time to perform some kind of maintenance or reconstruction. If this condition is further complicated by a relatively high use (high traffic) of the highway, the need for corrective action may become urgent.

Assuming data for these three parameters have been taken on the district highway network, the retrieval of the roughest, (low SI), or most slippery (low skid resistance coefficient) or high traffic (AADT) highway sections would be a relatively time consuming task if the records are in manual form. Compound this situation by requesting the retrieval of all pavement sections satisfying a specified constraint for each of the three parameters and the task becomes significantly large.

PFDS meets this need head-on. Figure 26 illustrates the instructions the District Engineer, for example, District 19, might give and the answers he would get from PFDS. He has specified that he wants a listing of all his pavements with an SI of 2.5 or below, a skid resistance coefficient of .35 or lower, and a traffic volume (AADT) greater than 500 vehicles per day. The computer output shows 14 sections that meet these conditions. This is a simple retrieval of the pavement sections identified by the record control key discussed in Chapter 4. Variation of the output format is achieved simply by adding the desired specifications to the request. For example, the District Engineer may have wanted the milepoint limits instead of or in addition to job number; a simple instruction to that effect would yield the desired information. The only constraint is that the requested parameter must be in at least one of the PFDS, skid resistance, or D-10 files.

*LIST: SI < 2.5 SKID-RESISTANCE < 0.35, AADT > 500, DISTRICT = 19.

*****	******	*****	******	******	******	*****	******	******
DISTRICT	CONTROL	SECT	J05 N0	ROADWAY	LANE	SI	SKID RESIST	AADT
****	****	*****	*****	******	******	*****	******	******
19	62	4	6	9	Ø	2.4	0.29	3013
19	62	ů,	8	0	ø	2.1	0.32	3013
19	62	4	9	0	21	2.1	0.33	3013
19	62	5	5	0	0	2.3	0.30	3013
19	62	5	7	Ø	Ø	2.5	0.29	3013
19	62	6	3	Ø	Ø	2.5	0.28	3013
19	62	6	4	7	Ø	2.4	0.28	3013
19	62	6	5	3	Ø	2.1	0.32	3013
19	62	7	7	Ø	Ø	2.2	0.33	3013
19	83	Å	2	0	Ø	2.0	0.31	774
19	83	8	3	0	Ø	2.3	0.31	774
19	83	9	5	0	2	2.2	0.33	774
19	83	9	5	3	- Ø	2.3	0.32	774
19	85	9	9	ð	Ø	2.4	0.30	774

Fig 26. Typical selective retrieval of pavement sections meeting specific conditions.

,

.

Information for Administration

Within the administrative user group are the Highway Commission, State Highway Engineer, and the headquarters divisions. Their needs for various kinds of information are often related to vital policy or programming actions and are therefore important in support of district efforts.

One relatively common activity is the review of pavement and highway designs accomplished by the districts. By combining their experience with all districts, the headquarters design review staff can help to preclude potential problems; they serve as a background upon which individual district problems can be projected to secure new, more reliable perspectives.

An example of the use of PFDS by the headquarters design review staff is illustrated in Fig 27. A district has submitted a set of plans specifying an asphalt-treated base, and the review staff wants to examine performance and location data on all pavements constructed with asphalt-treated bases in the past ten years. The constraints placed on the retrieval operation are those dictated by the particular design being reviewed. For the sake of simplicity, some important factors in pavement performance may not be explicitly shown and are assumed to be constant or not an influence in this case. Note that the PFDS output includes a computed value of serviceability loss per year. This type of data manipulation can be specified as desired. Once such analysis routines are written and used, they are catalogued and are available for use later simply by calling them by name.

The numbers reflected in this figure do not necessarily allow any conclusions about asphalt-treated base design. The real point of the illustration is that a tremendous amount of data has been screened to yield the desired information. If done manually, such an effort would entail months of effort; with PFDS, answers are available in minutes. Furthermore, no data are overlooked or forgotten as is frequently true with manual systems. The engineer is truly provided with the best possible feedback information to help him in his decision process.

Figures 28 and 29 illustrate the same retrieval operation in the MARK IV system. The initial complexity of the tabular format is quickly moderated after some brief study of the form. Training and experience with the system is needed to fully appreciate the simplicity and all of the advantages depicted here. For example, note that the four entries of the numeral one in

SWELLING	•CL4¥=P#08	48IF)1	Y < 3.85	*				
*****	******	*****	******	*******	*****	******	******	******
DISTRICT	CUNTRUL	SECT	JOH NO	ROADWAY	LANE	DATE	LAYER 2	SI LOSS
						CONSTRUCTED	THICKNESS	/ YEAR
******	*******	*****	******	*******	******	******	*******	******
(*2	314	3	3	ø	Ø	6810	8	0.21
¥15	297	5	7	୍ ପ	0	6206	8	55.0
26	1149	3	6	Ø	0	6608	10	0.17
29	258	8	16	3	0	6712	10	0,15
12	338	7	7	7	Ø	6808	7	0,25
14	573	5	5	0	0	6910	12	0.10
17	515	2	7	0	0	6682	8	0.19
14	523	4	i g	2	0)	6701	7	0.22
55	\$7	c	<u>21</u>	Ø	Ø	7006	14	0.05

*LIST SECTIONS ALL UINTHIETS: LAYER-1 = (ACP), THICKNESS = 3.0, LAYER-2 = (ASPHALT TREATED RASE), SWELL-RATE-CONSTANT < 0.04,

• • • • • • • • •

Fig 27. Sample retrieval of design data.

.





05/01/72	72 SAMPLE RETRIEVAL OF DESIGN DATA											
# # # * * * • •	DISTRICT	CONTROL	SECTION	JOB NO	ROADWAY	LANE	DATE Constr	LAYER Thick	SI Loss/yr			
	2	314		3			68/10	8	.21			
	5	297	5	7	Ø	Ø	62/06	8	.22			
	6	1140	3	6	Ø	Ø	67/12	10	.17			
	9	258	8	10	Ø	Ø	68/08	10	.15			
	12	338	3	7	0	Ø	69/10	7	,25			
	14	573	2	5	Ø	Ø	66/02	12	.10			
	17	212	2	7	Ø	Ø	67/04	8	.19			
	19	520	4	9	Ø	Ø	70/06	7	•55			
	22	37	8	4	Ø	Ø	70/05	14	.05			
GRAND CC	UNT			9								
M	Χ.							14	.25			
MI								7	171			
A Y	· · · ·								•1/3			

the three right columns produces the maximum, minimum, and mean values of the parameters designated (see Fig 29). It is difficult to conceive a more trivial programming task.

Information for the Research Engineer

The specific interests of the research engineer are served by PFDS as well. Just as the other groups of users can be assured of comprehensive data screening, the researcher can be confident that he has access to all the data collected. He need not fret that someone's "black book" was overlooked in his specific study area.

An example from the research realm might be a data retrieval and analysis operation to confirm (or deny) the theory that pavements exhibiting equivalent surface curvature indices (SCI) are equal in their traffic carrying capacity, regardless of their individual structural configuration. Figure 30 shows the PFDS output of District 15 data on performance of pavements with treated and untreated base courses. The analysis period is the first performance period only, i.e., to the first overlay. Data entries under the SCI levels are cumulative number of 18-kip single axle equivalents at time of overlay. It is assumed here that the overlays are performed when structural failure due to traffic has occurred; the constraints on swelling clay are intended to eliminate the factor from consideration in this problem.

The amount of data shown here are not sufficient to substantiate any firm conclusions, but there is an indication that the SCI theory may be invalid. For example, the traffic at failure for pavements with untreated bases, SCI range of 16 to 20 mils, varies from a low of 980,000 to a high of 2,030,000 18-KSA. Pavements with treated (cement) bases exhibit the expected lower SCI values but tend to require an overlay after fewer cumulative traffic applications. Again, the point is that PFDS not only contains the data needed to check and verify models but may provide the mechanisms to screen and manipulate data in any form desired by the researchers. The result should be greatly improved research efficiency and more research for each dollar invested. REQUEST NO.571: LIST PAVEMENT SECTIONS, DISTRICT FO 15, TRAFFIC EQ NUMBER 18 KSA *10**6 AT FIPST DVERLAY, RANGE OF SCI LEVELS EQ 5 MILS, SWELLING CLAY PROBABILITY LESS THAN 0.25 AND SWELL RATE CONSTANT LESS THAN 0.04, UNTREATED BASE (SPEC. NO. 232 503) THICKNESS GREATER THAN 7 INCHES AND LESS THAN 13 INCHES

SAME AS REQUEST DU-571, TREATED BASE (SPEC. NO.274 511) THICKNESS SAME AS 571.

REQUEST NO.572: LIST PAVEMENT SECTIONS. DISTRICT, TRAFFIC, SCI, SWELLING CLAY

REQUEST NO. 571 (UNTREATED FASE):

DISTRICT	CONTROL	SECT	JOB NO	ROADWAY	LANE	BASE	SCI LEVELS			
						THICKNESS	0=5 6=10 11=15 16=20 21=25			
15	17	6	2	Ø	0	10	1,55			
15	143	6	3	ø	Ø	8	1.82			
15	328	4	8	0	Ø	8	1.95			
15	470	3	5	2	Ø	10	0,98			
15	1740	1	4	6	e	12	2.03			

REQUEST NO. 572 (TREATED BASE):

15	72	7	14	ø	Ø	8	1.21
15	72	5	13	Ø	Ø	8	1.13
15	72	6	15	ø	Ø	10	1.25
15	142	14	17	Ø	Ø	10	1.49

Fig 30. Sample retrieval of research data.

.
This page replaces an intentionally blank page in the original --- CTR Library Digitization Team

CHAPTER 10. PFDS IMPLEMENTATION STRATEGY

The three major phases or stages of PFDS implementation are defined as

- (1) administrative decisions and actions,
- (2) pilot implementation, and
- (3) revision and extended implementation.

Each of these stages is composed of several distinct steps or sub-stages, and these are discussed in detail in the appropriate following sections.

Administrative Decisions and Actions

This grouping of specific steps should be considered priority actions generally in the order discussed.

- (1) Take immediate action to interface the skid resistance, maintenance, and proposed pavement feedback data systems. At the present time, D-8R has an operating skid resistance data system for Districts 14 and 19; D-18 is formulating a maintenance data system in Districts 19 and 21. Interaction of the responsible sections is imperative to insure that development efforts finally yield compatible but non-duplicative data systems.
- (2) In coordination with D-8, D-10, D-18, and D-19, administratively designate the record control key to be used in all automated highway system data files. In addition to the three systems mentioned above, the D-10 data files RL-1, Traffic Log and RI-2 must be standardized with the same record control key.
- (3) Proceed with implementation of the MARK IV data handling system as quickly as possible. This action should include:
 - (a) Training of D-8, D-10, D-18, and selected district personnel.
 - (b) Definition in MARK IV of PFDS files as now conceived.
 - (c) Redefinition in MARK IV of D-8R skid resistance files.
 - (d) Redefinition in MARK IV of RL-1, Traffic Log, and RI-2.
 - (e) Pilot runs of selected data file manipulations to check file and field compatibility.
- (4) Procure the additional MARK IV features
 - (a) Indexed Coordinated Files, to permit random access to the data base, and

- (b) Extended File Processing, to allow increased multiple file processing.
- (5) Accomplish initial coordination between D-8 and D-10 with the objectives of
 - (a) eliminating isolated problems with existing milepoint designations,
 - (b) beginning the inclusion of milepoints in RL-1, and
 - (c) revising design procedures to simplify later posting of pavement projects to RL-1 records.
- (6) Select one or more districts to effect a pilot implementation of PFDS. This action should be guided by the trial implementation experience with FPS, the skid resistance data files and the maintenance data system now being formulated. Specifically, the districts to test PFDS should be selected from the following summary of the districts now working with the indicated design or data system:

<u>FPS-11</u>	<u>Skid Resistance</u>	<u>Maintenance</u>
1	9	19
2	14	21
5	19	
8		
11		
14		
15		
17		
19		
21		

The selection of district(s) should also consider availability of special equipment such as a Mays Road Meter. While no single district may be initially capable or desirous of pilot implementation of all PFDS files, enough districts should be selected to permit total coverage of the PFDS files (each file being pilot implemented by at least one district).

Pilot Implementation

This phase should consist of the following specific steps:

 Conduct initial PFDS orientation session for selected districts. This should include an overview of PFDS concepts and its intended interaction with the applicable D-10 data files, skid resistance data system, and maintenance data system.

- (2) Immediately incorporate the essential factors specified by the selected districts and define/redefine the PFDS data files in MARK IV accordingly. Selected districts(s) should review the Traffic Log, RL-1, and RI-2 with a view to evaluating acceptability of current data and/or pinpointing additional changes.
- (3) Districts D-8 and D-10 should collaborate in development of some basic draft sampling plans to acquire the PFDS data on their respective interstate system. One or both of the research institutions should probably render assistance. This step should also include an inventory of existing district data acquisition procedures to
 - (a) evaluate suitability of existing data,
 - (b) effect necessary changes to existing data acquisition procedures,
 - (c) isolate new data collection requirements, and
 - (d) identify the logical district activity to carry out each PFDS data acquisition and reporting function.
- (4) Begin data collection and insertion in MARK IV files for the interstate highway system. Priority should be given to those sections designed with FPS-11.
- (5) Perform some basic data file manipulations to test interactive characteristics of all PFDS and related files (RL-1, RI-2, etc.)
- (6) Convene a workshop to include pilot districts and D-8, D-10, D-18, and D-19, to evaluate the results of pilot implementation.

Revision and Extended Implementation

The preceding pilot implementation of PFDS should yield some very valuable experience, thus making possible a first major revision to PFDS and associated procedures, and permitting a markedly increased implementation participation. Steps in this next phase are:

- Development and publication of PFDS instruction manuals. This must be a joint activity of the pilot districts, D-8, D-10, D-18, and D-19. Again, the Texas Transportation Institute and/or Center for Highway Research could provide assistance in writing these manuals.
- (2) Select additional districts to implement the full-scale PFDS. These districts should probably be those remaining on the FPS list shown previously. Conduct an orientation and training session for the new districts using manuals derived from the first step above.
- (3) After a three-month trial period by the new districts, convene a feedback workshop to review findings. From this collection of experience, revise PFDS files and procedures as appropriate. Some

"customizing" of PFDS files by districts may be warranted to satisfy unique situations.

- (4) Begin collecting data on the highway system in the following order, commensurate with resources:
 - (a) Selected FPS-designed pavements based on certain criteria
 - (b) Interstate system
 - (c) Remainder of system in order of traffic volumes.
- (5) Conduct feedback workshop and rewrite instruction and operating manuals accordingly.

Implementation Summary

Several important features of this proposed implementation plan are worthy of special comment:

(1) No additional district or division personnel are envisioned as a requirement to implement PFDS. The existing district staffs will be trained to properly input data to and retrieve data from PFDS and the related files. The MARK IV training sessions now being conducted by D-19 should adequately meet these needs. Personnel now responsible for data collection should be able to satisfy PFDS needs by intelligent evaluation of current procedures and reallocation of resources to only valid requirements. In other words, close scrutiny of existing data sampling and usage will undoubtedly reveal that some of this effort can be diverted to satisfy PFDS needs with no adverse effects.

In the headquarters divisions, the effective implementation of PFDS will depend upon a reordering of existing priorities. It is suggested that currently available personnel resources are adequate to support the PFDS implementation plan if a realistic appraisal of potential and relative benefits is made.

- (2) An open-ended iteration procedure for PFDS implementation is envisioned. Each loop expands upon the findings of the preceding one with a progressive increase of complexities. The pilot implementation phase can thus be considered as setting up a pattern for the succeeding phases. Figure 31 depicts a composite iteration of PFDS implementation.
- (3) An additional and immediate expenditure of \$10,000 is needed for the two additional MARK IV features. This sum may be reduced if some other agency of state government procures the features before the THD. Other costs will be incurred for measuring equipment on a progressive, phased basis, but no estimate of such costs can be defined at this time.



Fig 31. Composite iteration of PFDS implementation.

This page replaces an intentionally blank page in the original --- CTR Library Digitization Team

CHAPTER 11. CONCLUSIONS AND RECOMMENDATIONS

Conclusions

An operational PFDS for flexible pavement systems is dependent upon the enthusiastic support and action of some user districts. The selection of one or more districts to pilot test the system will serve as excellent implementation and allow further refinement based on a wealth of experience and a variety of viewpoints. However, care must be taken to prevent potential data pollution discussed many times throughout this report.

The MARK IV file handling system provides THD with a powerful tool that should generate benefits for all divisions and districts. This common file handling environment may encourage a wider dissemination of vital highway information that may generate interest in more record automation thus increasing the information potential in the Department. The pay-off should be a more informed Department and a continually improving highway system.

Recommendations

- (1) Begin immediately with PFDS implementation generally as outlined in Chapter 10.
- (2) Perform appropriate research needed to define essential factors of maintenance; insert appropriately in PFDS as a special research file until verified by performance feedback.
- (3) Begin identification of essential factors for rigid pavement design and management.
- (4) Consider automation of the control section records in D-8.
- (5) Evaluate need for a data files committee in THD to meet regularly to exchange information and eliminate data duplication.
- (6) Using NWS temperature records, perform some within-district computations for temperature constant to ascertain whether localized values of this parameter would contribute to improvement in reliability of the FPS design procedure.

This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

REFERENCES

- 1. AASHO Committee on Design, "AASHO Interim Guide for Design of Flexible Pavement Structures," Highway Research Board, April 1962.
- 2. Altman, James W., et al, "Handbook of Methods for Information Systems Analysts and Designers - Volume I," The Synectics Corporation, February 1971, NTIS" Document No. AD 725782.
- 3. American Association of State Highway Officials, "Proceedings, Committee on Computer Technology, National Conference," Albany, N. Y., May 1970, NTIS Document No. PB 204511.
- Blessing, William E., "Coordinated Data System for Highway Planning," <u>Highway Planning Technical Report No. 7</u>, May 1968.
- Brown, James L., and Hugo E. Orellana, "An Asphaltic Concrete Overlay Design Subsystem," Texas Highway Department Research Report No. 101-1F, December 1970.
- 6. Bureau of Program Planning, <u>HNDI System Planning Manual</u>, Wisconsin Department of Transportation, December 1, 1970.
- Byrnes, Carolyn J., and Donald B. Steig, 'File Management Systems: A Current Summary," <u>Datamation</u>, November 1969, pp 138-142.
- Carey, W. N., Jr., and P. E. Irick, "The Pavement Serviceability-Performance Concept," <u>Bulletin 250</u>, Highway Research Board, January 1960.
- 9. Clapp, Lewis, "Interactive Programming Systems and Languages," Computer Research Corporation, July 1971, NTIS Document No. AD 728224.
- Cochran, William G., <u>Sampling Techniques</u>, John Wiley and Sons, Inc., New York, 1963.
- Committee on Theory of Pavement Design, "Review of Existing Theories and Methods of Pavement Design," <u>Highway Research Circular No. 112</u>, October 1970.
- 12. Corbato, F. J., "Sensitive Issues in the Design of Multi-Use Systems," Massachusetts Institute of Technology, December 12, 1968, NTIS Document No. AD 688560.

[&]quot;NTIS - National Technical Information Service, U. S. Department of Commerce.

- 13. Darter, Michael I., "Applying Reliability Concepts to Pavement Design Systems," PhD Dissertation, The University of Texas at Austin, in progress.
- 14. Darter, Michael I., B. Frank McCullough, and James L. Brown, "Reliability Concepts Applied to the Texas Flexible Pavement System-FPS," a paper presented at 51st Annual Meeting of the Highway Research Board, January 1972.
- 15. Darter, Michael I., and James L. Brown, "Probabilistic Design Concepts Input for FPS-7," Technical Memo 123-18, Center for Highway Research, The University of Texas at Austin, June 17, 1971.
- 16. Dearden, John, F. Warren McFarlan, and William M. Zani, <u>Managing Computer-Based Information Systems</u>, Richard D. Irwin, Inc., Homewood, Illinois, 1971.
- 17. Delaney, William A., "Predicting the Costs of Computer Programs," <u>Data</u> <u>Processing Magazine</u>, October 1966, pp 32-34.
- 18. Dinnerstein, Lawrence I., "Compatibility Plus Enhancement The IBM System 370," Auerbach Info Inc., <u>Data Processing Magazine</u>, October 1970, pp 37-42.
- Durbin, E. P., "Information System Design in Large Scale Logistic Systems," The RAND Corporation, September 1970, NTIS Document No. AD 712693.
- 20. Fetter, Robert B., and James H. Carlisle, "Man-Computer Interaction in a Decision-Making Environment," Yale University, March 1971 (Report No. 1), NTIS Document No. 722336.
- Fry, J. P., et al, 'Data Management Systems Survey," The MITRE Corporation, January 1969, NTIS Document No. 684707.
- 22. Fry, J., and J. Gosden, "Survey of Management Information Systems and Their Languages," The MITRE Corporation, May 1968, NTIS Document No. AD 684706.
- 23. Gallo, Joseph C., and Donald J. Schilling, "Computer Programs to Detect Collusive Bidding on Public Contracts," Missouri Cooperative Highway Research Program Report 69-12, in cooperation with the U. S. Department of Transportation, NTIS Document No. PB 189322.
- 24. Glore, John B., "Major Problems of Generalized Data Management System Development," The MITRE Corporation, February 1970, M70-56.
- 25. Gosden, J., and E. Raichelson, "The New Role of Management Information Systems," The MITRE Corporation, April 1969, NTIS Document No. AD 691834.
- 26. Granley, Edwin C., "Quality Assurance in Highway Construction," <u>Public</u> <u>Roads</u>, Vol 35, No. 9, August 1969.

- 27. Greer, D. C., 'Research $1\frac{1}{\epsilon}$ Percent Funds," Interoffice Memorandum to Mr. Dingwall, Texas Highway Department, October 23, 1962.
- 28. Haas, R. C. G., 'Developing a Pavement Feedback Data System," Research Report No. 123-4, published jointly by the Texas Highway Department; Center for Highway Research, The University of Texas at Austin; and Texas Transportation Institute, Texas A&M University, February 1971.
- 29. Haas, R. C. G., "Some Introductory Concepts for a Pavement Feedback Data System," Technical Memo 123-06, Project 123, Center for Highway Research, The University of Texas at Austin, June 8, 1970.
- 30. Haas, R. C. G., W. Ronald Hudson, B. F. McCullough, and James L. Brown, "Developing a Pavement Feedback Data System," a paper presented to the annual meeting of Highway Research Board, Washington D. C., January 1972.
- 31. Haas, R. C. G., and W. R. Hudson, "The Importance of Rational and Compatible Pavement Performance Evaluation," prepared for presentation at the Highway Research Board Western Summer Meeting, Sacramento, California, August 17-19, 1970.
- 32. Head, Robert V., 'Management Information Systems: A Critical Appraisal," Datamation, May 1967, pp 22-27.
- 33. Ilead, Robert V., "The Elusive M.I.S.," <u>Datamation</u>, September 1, 1970, pp 22-27.
- 34. Highway Network Data and Information (HNDI) Section, "System Continuance Report," Wisconsin Department of Transportation, Madison, Wisconsin.
- 35. Highway Research Board, 'Development of Wisconsin's Integrated Operation System," Highway Research Record No. 326, 1970.
- 36. Highway Research Board, "Quality Assurance and Acceptance Procedures," <u>Highway Research Board Special Report 118</u>, 1971, pp 17-22.
- 37. Highway Research Board, "Structural Design of Asphalt Concrete Pavement Systems," Proceedings of Workshop held December 7-10, 1970, Austin, Texas, (Preliminary Draft).
- 38. Hudson, W. R., F. N. Finn, B. F. McCullough, K. Nair, and B. A. Vallerga, "Systems Approach to Pavement Design," Interim Report NCHRP Project 1-10, submitted to National Cooperative Highway Research Program, Highway Research Board, March 1968.
- 39. Hudson, W. R., B. F. McCullough, and Fred N. Finn, 'Factors Affecting Performance of Pavement Systems," <u>Transportation Engineering Journal</u>, Vol 95, No. TE3, American Society of Civil Engineering, August 1969, pp 505-519.

- 40. Hudson, W. Ronald, B. Frank McCullough, F. H. Scrivner, and James L. Brown, "A Systems Approach Applied to Pavement Design and Research," Research Report 123-1, published jointly by the Texas Highway Department; Center for Highway Research, The University of Texas at Austin; and Texas Transportation Institute, Texas A&M University, March 1970.
- Hutchinson, B. G., and R. C. G. Haas, "A System Analysis of the Pavement Design Process," <u>Highway Research Record No. 239</u>, Highway Research Board, 1968.
- 42. IBM Corporation, 'Data File Handbook," GC 20-1638-2 Third Edition, September 1969.
- 43. Informatics, Inc., 'MARK IV File Management System Can You Afford Not to Have It?," Sales Brochure, 1971.
- 44. Informatics, Inc., "MARK IV File Management System User's Guide," Document No. SP-69-810-3, July 1969.
- 45. Jain, Surendra Prakash, B. Frank McCullough, and W. Ronald Hudson, "Flexible Pavement System - Second Generation, Incorporating Fatigue and Stochastic Concepts," Report No. 123-10, published jointly by the Texas Highway Department; Center for Highway Research, The University of Texas at Austin; and Texas Transportation Institute, Texas A&M University, December 1971.
- 46. Jones, M. V., "Computerizing a Government Data System: A Management Overview of the Steps Required and the Time Needed," The MITRE Corporation, July 1969, NTIS Document No. AD 692784.
- 47. Kher, Ramesh K., B. Frank McCullough, and W. Ronald Hudson, "A Sensitivity Analysis of Flexible Pavement System FPS 2," Research Report 123-8, published jointly by the Texas Highway Department; Center for Highway Research, The University of Texas at Austin; and Texas Transportation Institute, Texas A&M University, August 1971.
- 48. Konder, Robert L., and Raymond J. Krizek, "Factors Influencing Flexible Pavement Performance," NCHRP Report 22, Highway Research Board, 1966.
- 49. Kreger, Alan, and Janet Nathanson, "The Tribulations and Triumphs of GIS," <u>Datamation</u>, October 15, 1971, pp 20-25.
- 50. Kriebel, Charles H., "The Evaluation of Management Information Systems," Carnegie-Mellon University, September 1970, NTIS Document No. AD 723083.
- 51. Kriebel, Charles H., "Perspectives on Information Processing in Management Information Systems," Carnegie-Mellon University, August 1969, NTIS Document No. AD 695767.
- 52. Lemer, A. C., and Fred Moavenzadeh, "An Integrated Approach to Analysis and Design of Pavement Structure," <u>Highway Research Record No. 291</u>, Highway Research Board, 1969, pp 173-185.

- 53. Lytton, R. L., "Prediction of Pavement Roughness Using Expansive Clay Theory," unpublished notes of study at Texas Transportation Institute, June 1971.
- 54. MacGowan, Roger A., "Computers and Data Processing Expert Opinion Survey," Department of Defense Computer Institute, June 1968, NTIS Document No. AD 851541.
- 55. Madill, J. P., and J. Kuss, "Study and Recommendations for Acquisition of Generalized Data Management Systems," Report No. 3, Simon-Fraser University, British Columbia, February 1971.
- 56. McCullough, B. Frank, James L. Brown, W. Ronald Hudson, and F. H. Scrivner, "Annual Report on Important 1970-71 Pavement Research Needs," Report No. 123-7, published jointly by the Texas Highway Department; Center for Highway Research, The University of Texas at Austin; and Texas Transportation Institute, Texas A&M University, April 1971.
- 57. McCullough, B. F., Ramesh Kher, and Phil Smith, "Identification of Pavement Distress Manifestations and Methods of Treatment," Study No. 3 8/18-72, Research Study Proposal to Texas Highway Department from the Center for Highway Research, The University of Texas at Austin, March 9, 1972.
- 58. MRI Systems Corporation, "System 2000," general information manual, Austin, Texas, 1971.
- 59. Peebles, Richard, 'Design Considerations for a Large Computing System with a Geographically Dispersed Demand," The Moore School of Electrical Engineering, March 1970, NTIS Document No. AD 702759.
- 60. Peterson, Dale E., W. Ronald Delis, and Dennis L. Mangrum, "A System for Planning Roadway Improvements," Materials and Test Division, Utah State Department of Highways.
- 61. Pister, Karl S., "Some Remarks on Research for Structural Design of Asphalt Concrete Pavement Systems," a paper presented at Workshop on Structural Design of Asphalt Concrete Pavement Systems, Highway Research Board, Austin, Texas, December 1970.
- 62. "Placing Milepost Markers," letter from State Highway Engineer to District Engineers, Texas Highway Department, May 6, 1968.
- 63. Schlessinger, Bernard S., "Proceedings of the Scientific and Technical Information Seminar (5th) Held at Southern Connecticut State College, New Haven, Connecticut on April 9, 1970," March 8, 1971, NTIS Document No. AD 722463.
- 64. Schwartz, J. I., "Advanced Development Prototype Final Report (Part I) to the Director, Advanced Research Projects Agency for the Period July 30, 1968 to January 30, 1969," Systems Development Corporation, January 30, 1969, NTIS Document No. AD 683670.

- 65. Scrivner, F. H., and Chester H. Michalak, "Flexible Pavement Performance Related to Deflections, Axle Applications, Temperature, and Foundation Movements," Research Report 32-13, Texas Transporation Institute, Texas A&M University, College Station, Texas, 1969.
- 66. Scrivner, F. H., C. H. Michalak, and W. M. Moore, "Calculation of the Elastic Moduli of a Two Layer Pavement System from Measured Surface Deflections," Report No. 123-6, published jointly by the Texas Highway Department; Center for Highway Research, The University of Texas at Austin; and Texas Transportation Institute, Texas A&M University, March 1971.
- 67. Scrivner, F. H., and W. M. Moore, "An Empirical Equation for Predicting Pavement Deflections," Research Report 32-12, Texas Transportation Institute, Texas A&M University, College Station, Texas, 1968.
- 68. Scrivner, F. H., W. M. Moore, W. F. McFarland, and G. R. Carey, "A Systems Approach to the Flexible Pavement Design Problem," Research Report 32-11, Texas Transportation Institute, Texas A&M University, College Station, Texas, 1968.
- 69. Scrivner, F. H., and W. M. Moore, "Some Recent Findings in Flexible Pavement Deflections," Research Report 32-9, Texas Transportation Institute, Texas A&M University, College Station, Texas, 1967.
- 70. Shahin, Mohammed Y., "Prediction of Low Temperature and Thermal Fatigue Cracking in Flexible Pavements," PhD Dissertation, The University of Texas at Austin, in progress.
- 71. Sundeen, Donald H., "General Purpose Software," <u>Datamation</u>, January 1968, pp 22-27.
- 72. Texas Highway Department, <u>Construction Manual</u>, Construction Division, October 1966, Revised January 1971.
- 73. Texas Highway Department, <u>Flexible Pavement Designer's Manual, Part I</u>, March 1970.
- 74. Texas Highway Department, <u>Highway Design Division Operating and Pro-</u> cedures Manual, 1970 Edition.
- 75. Texas Highway Department, <u>Maintenance Manual</u>, Maintenance Operations Division.
- 76. Texas Highway Department, "Plan Preparation Book I," a manual prepared by Instructor's School, Texas Highway Department by the Road Design Division, December 1952.
- 77. Texas Highway Department, "Roadway Code Chart, Road Life Studies," Planning Survey Division, January 1970.

- 78. Texas Highway Department, "Systems Analysis of the Planning Survey Division," Vols 1 and 2, Staff report prepared in cooperation with the U. S. Department of Transportation, June 1, 1968.
- 79. Texas Water Development Board, "Committee Review of The University of Texas's Remote File Management System (RFMS)," Electronic Data Processing Division.
- 80. The Austin American, "\$200 Million Highway Skimp," Austin, Texas, July 31, 1970, p 4.
- 81. Utah State Highway Department, "Evaluation of Pavement Serviceability on Utah Highways," Interim Report 1969, Utah Project HPR 1 (8).
- 82. Vance, John A., 'Geographical Data Coding Grid,'' <u>Proceedings</u>, Canadian Good Roads Association, 1966, pp 265-278.
- 83. Vesic, Aleksander S., William H. Perloff, and Carl L. Monismith, editors, "Review of Existing Theories and Methods of Pavement Design," Circular Number 112, Highway Research Board, October 1970.
- 84. Walker, Roger S., Freddy L. Roberts, and W. Ronald Hudson, "A Profile Measuring, Recording, and Processing System," Research Report No. 73-2, Center for Highway Research, The University of Texas at Austin, April 1970.
- 85. Walker, Roger S., "Tentative Texas Highway Department Procedures for the Calibration, Operation and Control of the Mays Road Meter," Research Project 3-8-71-156, Center for Highway Research, The University of Texas at Austin, February 1972.
- 86. Wisconsin Department of Transportation, 'Data Needs Analysis," attachment to departmental correspondence, September 2, 1971.
- 87. Wisconsin Department of Transportation, "Highway Network Data and Information (HNDI) Report," a report of HNDI data retrieval examples.
- 88. Wisconsin Department of Transportation, "HNDI System Narrative," description of the Highway Network Data and Information (HNDI) System.
- 89. Wisconsin Department of Transportation, 'Requirements and Specifications for the Defense Bridge Reports," System Continuance Report, May 13, 1971.
- 90. Witt, Arthur W., III, and B. Frank McCullough, "Characterization of the Swelling Clay Parameter Used in the Pavement Design System," Research Report No. 123-3, published jointly by the Texas Highway Department; Center for Highway Research, The University of Texas at Austin; and the Texas Transportation Institute, Texas A&M University, August 1970.

- 91. Wright, J. E., "Developing a State-Wide Data Base in Texas," a paper presented at the Highway Research Board Symposium on Automation Systems for Highway Organizations, held at The University of Texas at Austin, August 16-18, 1971.
- 92. Yu, Francis C. F., "An Alternate Key for a Statewide Highway Information Retrieval System for the State of Texas," unpublished course paper at The University of Texas at Austin, December 1971.
- 93. Yu, Francis C. F., "Design of a Computer-Based Information Retrieval System for Highway Planning," Master of Science Thesis, Montana State University, December 1970.
- 94. Yu, Francis C. F., "MARK IV," Technical Memo 123-25, Center for Highway Research, The University of Texas at Austin, March 29, 1972.

ABBREVIATIONS

٠

APPENDIX A

This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

ABBREVIATIONS

AADT	-	Annual average daily traffic.
AASHO	-	American Association of State Highway Officials.
ACP	-	Asphaltic concrete pavement.
COGENT III	-	A COBOL – compatible generalized file management system developed by Computer Sciences Corporation.
DM-1	-	Data Manager -l, a generalized file management system designed by Auerbach Corporation.
FPS	-	Flexible pavement system.
GDMS	-	Generalized data management system.
GIS	-	Generalized information system, a collection of programs to support formatted file functions, designed/written by International Business Machines (IBM).
HNDI	-	Highway Network Data and Information System, the Wisconsin Department of Transportation technical data system.
ISAM	-	Indexed sequential access method, a programming method that allows rapid access to file records in random fashion.
KSA	-	Kips single axles, used with numeral 18 to designate the common traffic base of 18 kip single axles.
MARK IV	-	An advanced general-purpose data management system developed by Informatics, Incorporated.
MIS	-	Management information system.
NIPS	-	National military command information processing system, a general purpose file handling system adapted to military in- telligence functions from IBM's formatted file system (FFS).
NTIS	-	National Technical Information Service, U. S. Department of Commerce, Springfield, Virginia 22151.
NWS	-	National Weather Service.
PFDS	-	Pavement feedback data system.
PL/1	-	Programming language l, an IBM "super" version of FORTRAN and COBOL computer programming languages.
PSI	-	Present serviceability index, now commonly referred to only as SI, serviceability index.
RI-2	-	Road inventory file maintained by the Planning Survey Division of Texas Highway Department.
RL-1	-	Road life file maintained by the Planning Survey Division of Texas Highway Department.

SCI	-	Surface curvature index.										
SI	-	Serviceability index, synonomous with PSI:										
THD	-	Texas	Texas llighway Department.									
		Heado	luar	ters Divisions:								
		D - 3	-	Finance								
		D - 4	-	Equipment and Procurement								
		D - 5	-	Bridge								
		D - 6	-	Construction								
		D - 8	-	Highway Design								
		D - 9	-	Materials and Tests								
		D -1 0	-	Planning Survey								
		D-18	-	Maintenance Operations								
		D-19	-	Automation								
WISDOT	-	Wisco	ons	in Department of Transportation.								

APPENDIX B

PFDS SPECIFICATIONS AND MACRO COMPARISON OF GDMS

This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

PFDS SPECIFICATIONS AND MACRO COMPARISON OF GDMS

	the second se	the second s				
GDMS Specifications for PFDS	DM-1	COGENT	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
MANDATORY IBM 360/50	360 <i> 5</i> 0	360/40	360/30	360/40	360/40	Y
Operating System (OS) IBM	OS	OS	OS, DOS	OS	os	Y
Vender and Cost	Auerbach Cerp.	Computer Sciences Corp.	Informatics Inc. \$33,000 +	IBM \$400/mo.	(Dept. of Defense)	Free Package + Programming
Index Sequential Organization	Y	Y	Y	Y	Y	Y
Hierarchical File with Multi-level Arrangement	У	Y	Y	Y	Y	Y

(Continued next page)

Legend

Y -Yes, compatible DOS-Disc operating system

GDMS Specifications for PFDS	DH-1	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
Update (add, delete, and modify)	Y	Y	Y	Y	Y	ISAM
Vendor Supplied IOCS	Y	Y	Y	Y	Y	Y
Scientific Library Subroutine Links	Y	Y	Y	Y	N	IBM Scientific Routines
DASD	Y	Y	Y	Y	Y	3336 2314
Procedural Language DO	+	N	EQ	Y	N	Function of Command Processor of PFDS

(Continued next page)

Legend

Y -Yes, compatible

- N -No
- -Need to implement

EQ -Equivalent

10CS-Input-Output control system

DASD-Direct access storage device

PFDS SPECIFICATIONS AND MACRO COMPARISON OF GDMS (Continued)

GDMS Specifications for PFDS	DM-1	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
GO TO	#	NM	BQ	Y	N	Function of Command Processor of PFDS
IF	+	NM	EQ	Y	N	Function of Command Processor of PFDS
Mode of Computation: Floating Point	+	Y	Y	N	N	Feature of PL/1
Decimal	+	Y	Y	Y	N	Feature of PL/1
Integer	+	Y	Y	N	Y	Feature of PL/1

(Continued next page)

Legend

-Need to implement

NM -Need medification

EQ -Equivalent

Y -Yes, compatible

N -No

GDMS Specifications for PFDS	DH-1.	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
Operators: = ± X / EXP	#	Y	Y	Y	NM	Feature of PL/1
Vlog Trigonometric Functions	#	*	#	N	#	Feature of PL/1
Logical and Boolean	Y	Y	Y	Y	Y	Feature of PL/1
Arithmetic Expression (number of operations per statement)	*	1	1	N	#	64
Built-in Engineering Analysis	N	N	N	N	N	Within Capability of PL/1

(Continued next page)

Legend

F -Need to implement Y -Yes, compatible NM -Need modification

NM -NGGA MOATIIG

N -No

PFDS SPECIFICATIONS AND MACRO COMPARISON OF GDMS (Centinued)

Specifications for PFDS	DH-1	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
Documentation (detailed system)	NF	NF	Proprietary	Y	Not Generally Available	#
File Reorganization to Improve Efficiency	÷	*	#	*	+	Read File Sequentially Onto New Storage Area Before Releasing Old Storing Area
Ease of Language Interface	Function of Chosen Language	COBOL	Restricted	Difficult	Difficult	PL/1
Missing Input Data Elements Permitted	Y	NF	N	N	NM	Feature of PL/1

(Continued next page)

Legend

NF -Not firm

Y

-Yes, compatible -Need to implement ŧ

-No N

NM -Need modification

and the second		and the second		and a strength of the second s	and the second state of th	
GDMS Specifications for PFDS	DM1.	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
Security: 1. File	¥	Y	+	Y	Ŷ	JCL
2. Entry, Greap, Stc.	¥	Y	*	N	N	JCL
DESIRABLE						
Data Element Physical Size: 1. Variable Length	2 <i>5</i> 4B		N	N	NM	255B
2. Fixed Length a, Numeric	2 <i>5</i> 4B		4B	31 DD	4B	31 DD
b. Alphanumeric	2 54B		255B	255 B	255B	255B
3. Variable Name	#	2 9	8			31

(Continued next page)

Legend

Y -Yes, compatible

-Need to implement

N -No

NM -Need modification

Blank means unknown.

JCL -Job control language B -Bytes DD -Data digits

PFDS SPECIFICATIONS AND MACRO COMPARISON OF GDMS (Continued)

The second s						
GDMS Specifications for PFDS	DM-1	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
Levels of Nesting			9	4	8	64
Temperary Hold File	Ŷ	Y	Ŷ	Y	NM	SYSOUT (JCL)
Update Multiple Files in One Pass	#	Y	Y	NM	N	#
Flexible Data Elements (overriding data attributes)	+	Y	N	Y	Y	Dynamic Declaration of Data Structure in Program
Data Name Directed Update	#	Y	N	N	Y	PL/1

(Continued next page)

Legend

Y -Yes, compatible NM -Need modification # -Need to implement N -No Blank means unknown.

GDMS Specifications for PFDS	DM-1	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
List Directed Update	#	Y	Y	Y	Y	PL/1
Interruption and Resovery			Y			"ON" Options of PL/1
Input Edit (error check): 1. Maximum or Minimum Values	*	Y	Y	NM	Y	As Programmed
2, Range of Values	#	Y	Y	Y	Y	As Programmed
3. Specific Characters	+	Y	Y	Y	Y	As Programmed
4. Sequency or Identify	#	Y	Y	Y	Y	As Programmed
5. Cross Comparison	#	Y	Y	NM	Y	As Programmed

(Continued next page)

Legend

-Need to implement

Y -Yes, compatible

NM -Need modification

PFDS SPECIFICATIONS AND MACRO COMPARISON OF GDMS (Continued)

				· · · · · ·		
GDMS Specifications for PFDS	DM-1	COGENT	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
Decode and Encode	NF	Y	#	Y	У	As Programmed
Subordinate File Crea- tion from Source File	+	Y	Y	N	N	Derived from Source File from Programming
Physical Format of the Input Data to Generate a File: 1. Must be Specific	+	NF	N	N	N	Not Necessary
2. May be Several	#	NF	Y	Y	Y	Optional
3. May be Any	#	NF	NM	Y	N	As Programmed
Pagination	ŧ	Y	Y	Y	Y	As Programmed

(Continued next page)

Legend

NF -Not firm

Y -Yes, compatible

-Need to implement

N -No

NM -Need modification

Concernent and the second seco		14	has a set of the	And a state of the second s		
GDMS Specifications for PFDS	DM-1	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
Size, Title, Line, Positioning, etc.	*	Y	Y	У	У	As Programmed
Statistical Functions	ŧ	NF	N	N	Y	IBM Routines
Picture Specification (input/output)	ŧ	Y	Y	N		PL/1
Backup Capability	#					Utilities
Dynamic Storage in Programming						AUTOMATIC, CONTROLLED, BASED, ALLOCATE, and FREE

(Continued next page)

Legend

-Need to implement Y -Yes, compatible

NF -Not firm

N -No

Blank means unknown.

PFDS SPECIFICATIONS AND MACRO COMPARISON OF GDMS (Continued)

GDMS Specifications for PFDS	DM-1	COGENT III	MARK IV	GIS	NIPS	PFDS Using ISAM + PL/1 + JCL + Utilities
OPTIONAL Online Configuration	#	NF	NF	Y	Y	THD Facility
Multiple Console/ Terminals	ŧ	NF	NF	Y	Y	THD Facility
Interactive Mode (tutorial)	*	NF	NF	N	N	As Programmed
System Tallies (keeping track)	+	NF	Y	Y	Y	os
Sorting Capability	#	Y	Y	Y	Y	Utilities

Legend

_Need to implement

NF _Not firm

Y _Yes, compatible

N **No**

OS -Operating system

This page replaces an intentionally blank page in the original --- CTR Library Digitization Team

APPENDIX C

TRAFFIC LOG CODING FORM AND SAMPLE FILE OUTPUT
This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

Job	Bumber			Fi (21	le Name TLOG nd Revisi	on)	Storage Medium Tape				
Avg. 1	lo. Recon	rds	Fi	elds 41	Char 1	racters 72	How Current 1970 Data	Date Prepared 5-20-71			
File S	Sequence		·				Prepared By B. B.	Reviewed 3;			
Labels	5							Record Source			
Retent	tion Chai	racteri	stics					Page 1 of 2			
Remark	(19)	Milepo 70 and	ints <u>will</u> later).	be out	of sort	in the TLOG	Master File				
Item No.	From To	Size	No. of Dec. Pos.	Field Char.	Lab el	-	Item Name				
. 1	1-2	2		I		DISTRICT NU	JMBER				
2	3-5	3		I		COUNTY NUME	BER				
3	6-9	4		I		CONTROL					
4	10-11	2		I		SECTION					
5	12-16	5	3	I		BEGINNING N	ILEPOINT				
6	17-21	5	3	I		ENDING MILE	CPOINT				
7	22-26	5	3	I		LENGTH OF S	SEGMENT				
8	27-30	4		I		HIGHWAY NUN	BER				
9	31-32	2		I		HIGHWAY SYS	STEM				
10	33-34	2		I		ADMINISTRAT	TIVE SYSTEM				
11	35-36	2		I		YEAR (OF CL	urrent ADT Field)	· · · · · · · · · · · · · · · · · · ·			
12	37-42	6		I		ADT (Current Year)					
13	43-48	6		I		ADT (Year-1	1)				
	49-54	6		I		ADT (Year-2					
14		6		I		ADT (Year-3	3)				
14 14 15	55-60		1	I		ADT (Year-4	+)				
14 15 16	55-60 61-66	6	┥ →			4070 / 17	··· · · · · · ·				
14 15 16 17	55-60 61-66 67-72	6 6		I	•·	ADI (lear-:) 	Historical ADI			
14 15 16 17 18	55-60 61-66 67-72 73-78	6 6 6		I I		ADT (Year-6	5)	Historical ADI Fields			
14 15 16 17 18 19	55-60 61-66 67-72 73-78 79-84	6 6 6 6			· · · · · · · · · · · · · · · · · · ·	ADT (Year-6 ADT (Year-6 ADT (Year-7	5) 5) 7)	Fields			

.

Jo	C.02	Ng		(2nd	le Name FLOG 1 Revisio	on)	Таре					
Avg.	No. Recor	rds	Fi 4	elds 1	Cha	racters 172	How Current 1970 Data	Date Prepared				
File	Sequence	Reviewed By										
Label	s		Record Source									
Reten	tion Char	acteris	stics					Page 2 of 2				
Remar	ks		-					······				
Item No.	From To	Size	No. of Dec. Pos.	Field Char.	Label		Item Name					
22	97-104	8	•	I		VEHICLE MILE	S, DAILY (Curren	nt Year)				
23	105-106	2		İ		ADMINISTRATI	VE SYSTEM					
24	107-108	2		I		YEAR						
25	109-112	4	2	I		INCREASE FAC	TOR (%)	Design Year				
26	113-118	6		I		ESTIMATED AD	Т					
27	119-126	8		I		ESTIMATED V.	M					
28	127-129	3	1	I		"K" FACTOR		· · ·				
29	130-131	2		I		DIR. DIST.						
30	132-134	3	1	I		% TRUCKS IN	AADT					
31	135-137	3	1	I		& TRUCKS IN	D.H.V.	Traffic Data				
	138-140	3		I		ATHWLD (Hund	ired Lbs.)	20 Year				
32	141-142	2		I		ATHWLD (% Ta	ndem Axles)	Design Period				
32 33		5		I		TNWLGT 8000	Lbs. (K)					
32 33 34	143-147		l i	I		FLEX. (K)	Analysis					
32 33 34 35	143-147 148-152	5				PTCTD (K)						
32 33 34 35 36	143-147 148-152 153-157	5 5		I			F.C.					
32 33 34 35 36 37	143-147 148-152 153-157 158	5 5 1		I I		F.C.						
32 33 34 35 36 37 38	143-147 148-152 153-157 158 159	5 5 1 1		I I I		F.C. URBAN-RURAL	Functional (Classification				
32 33 34 35 36 37 38 39	143-147 148-152 153-157 158 159 160	5 5 1 1 1		I I I I	· · · · · · · · · · · · · · · · · · ·	F.C. URBAN-RURAL CONNECT LINK	Functional (Classification				
32 33 34 35 36 37 38 39 40	143-147 148-152 153-157 158 159 160 161-166	5 5 1 1 1 6		I I I I I	· · · · · · · · · · · · · · · · · · ·	F.C. URBAN-RURAL CONNECT LINK SERIAL NUMBE	Functional (C	Classification				

STRICT - INTY - 1	5 52	MAR 1. Lubbock	1971	PLANN	IING SURA	VEY DIN	ISION	- 1969 TF	RAFFIC	LOG		PAGE -	008 (A)
SERIAL ND+	NEXT YEAR	CONTROL SECTION	MILEPDIN BEGIN - I	TS END LENG	HIGHWAY NO.SYST	ADM. SYST	R U 1969	- HISTOR 1968	RICAL A	VERAGE DA	962	1961	1960
01620	÷.`	67 15	616 10	583 1067	461 7	3 1	R 1120	1260	0	0/		0	0
01630		68 1	0	149 149	87 1	2	1 15305	16195	ŏ	ő/		õ	õ
01640		68 1	149	563 414	87 1	2	U 15735	16615	· õ	ő/		ŏ	0
01550		68 1	563 12	238 675	87 1	Z	U 12297	14105	. 0	o/	\ [°] õ	0	0
01660		68 l	1238 2	211 973	87 1	1	R 8040	10890	. 0	1	0	0	0
01580		68 1	2211 32	210 999	87 1	1	R 6520	6500	0	1	/ 0	0	0
01690		68 1	3210 42	211 1001	87 1	1	R 5880	5890	0	\mathcal{O}	6 0	0	0
01700		63 1	4211 4	710 499	87 1	1	R 4380	4350	0	12	0 0	0	0
01710		68 1	4710 41	830 120	87 1	1	R 4390	4350	0		/0 0	0	0
01720		68 1	4830 49	960 130	87 1	1	R 4420	4330	0	/ /	0 O	0	0
01730		68 1	4960 52	212 252	87 1	1	R 4360	4330	0	1 1	0 0	0	0
01740	0	68 l	5212 5	812 600	87 1	. 1	R 4230	4300	0		0 0		0
01760		68 1	6812 7.	336 524	87 1		R 4120	4080	0		0 0		0
TRICT -	/ 	MAR 1.	1971	PI ANN			196	59 TRAFFI					
TRICT - NTY - 1	5 52	MAR 1, LUBBOCK	1971	PLANN	ING SURV	VEN T	196	59 TRAFFI	C LOG	D TRAFFI	C DATA		P
TRICT - NTY - 1	52	MAR 1. LUBBOCK	1971	PLANN GN YEAR	ING SURV		196	59 TRAFFI % TRUCKS	C LOG RELATE	D TRAFFI	C DATA	PAVE	P 1ENT
TRICT - NTY - 1 SERIAL	5 52 1969	MAR 1. LUBBOCK	1971 DESI(PLANN GN YEAR	ING SURV		196 DIR	59 TRAFFI % TRUCKS IN IN	C LOG RELATE	ED TRAFFI HWLD 8	C DATA TNWLGT	PAVE	P MENT YSIS
TRICT - NTY - 1 SERIAL NO.	5 52 1969 VEH MIL	MAR 1, LUBBOCK 	1971 DESIG R FAC	PLANN GN YEAR ~- ESTIM	ING SURA	VEV T FAC	DIR DIST	59 TRAFFI % TRUCKS IN IN ADT DHV	C LOG RELATE AT	ED TRAFFI HWLD % TANDEM	C DATA TNWLGT 8000 LBS	PAVE ANALY FLEX	P IENT YSIS RIDG
TRICT - NTY - 1 SERIAL NO.	5 52 1969 VEH MIL	MAR 1. LUBBOCK .E ADM Y	1971 DESI(R FAC (XX.XX)	PLANN GN YEAR ~- ESTIM ADT	ATED	FAC	DIR DIST	59 TRAFFI % TRUCKS IN IN ADT DHV { XX+X }	C LOG RELATE AT 100 LBS	D TRAFFI HWLD % TANDEM AXLES	C DATA TNWLGT 8000 LBS (K)	PAVE ANALY FLEX	P IENT YSIS RIDG
TRICT - NTY - 1 SERIAL NO. 01620	5 52 1969 VEH MIL	MAR 1, LUBBOCK .E ADM Y 25 0 9	1971 DESI(R FAC (XX.XX) 0 600	PLANN GN YEAR ESTIM ADT 2530	ATED V.M. 279	УЕЧ Г К FAC 10	196 DIR DIST 54	59 TRAFFI % TRUCKS IN IN ADT DHV { XX.X } 162 118	C LOG RELATE AT 100 LBS 97	ED TRAFFI HWLD % TANDEM AXLES 20 20	C DATA TNWLGT 8000 LBS (K) 211	PAVE ANALY FLEX 465	P MENT YSIS RIDG 555
TRICT - NTY - 1 SERIAL NO. 01620 01640	5 52 1969 VEH MIL 110 226	MAR 1, LUBBOCK .E ADM Y 95 0 9	1971 DESI(R FAC (XX.XX) 0 600 540	PLANN GN YEAR ESTIM ADT 2530 32660	ATED V.M. 270 49	УЕЧ Г К FAC 10-	196 DIR DIST 54 54	59 TRAFFI % TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56	C LOG RELATE AT 100 LBS 97 105	ED TRAFFI HWLD % TANDEM AXLES 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369	PAVE ANALY FLEX 465 3046	P MENT YSIS RIDG 555 3617 2444
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01650	5 52 1969 VEH MIL 110 226 651	MAR 1, LUBBOCK .E ADM Y 95 0 9 10 0 9	1971 DESI((XX.XX) 0 600 540 510	PLANN GN YEAR ESTIM ADT 2530 32660 32590 32610	ATED V.M. 279 49 17	УЕУ Г К FAC 10 100	196 DIR DIST 54 54 54	59 TRAFF I % TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56 79 55	C LOG RELATE AT 100 LBS 97 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379	PAVE ANALY FLEX 465 3046 3069	P MENT (SIS RIDG 555 3617 3644 3794
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01660	5 52 1969 VEH MIL 110 226 651 830 782	MAR 1, LUBBOCK .E ADM Y 95 0 9 10 0 9 14 0 9 10 0 9	DESI R FAC (XX.XX) D 600 D 540 D 510 D 860 D 800	PLANN GN YEAR ESTIM ADT 2530 32660 32590 34510 21550	ATED V.M. 270 49 17	УЕУ Г К FAC 10 10 10 10	DIR DIR DIST 54 54 54 54	59 TRAFF I 7 TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56 79 55 85 60 101 71	C LOG RELATE AT 100 LBS 97 105 105	D TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079	PAVE ANALY FLEX 465 3046 3069 3194 2302	P MENT (SIS RIDG 555 3617 3644 3794 2846
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01650 01660 01680	5 52 1969 VEH MIL 110 226 651 830 782 651	MAR 1, LUBBOCK .E ADM Y 95 0 9 90 0 9 90 0 9 90 0 9 90 0 9 3 0 9	1971 DESI((XX.XX) 0 600 0 540 0 510 0 860 0 800 0 1120	PLANN GN YEAR ESTIM AOT 2530 32660 32590 34510 21550 21860	A TED V. M. 2 TO 4 P 17	ИЕЧ Г К FAC 10 10 10 10 10	DIR DIST 54 54 54 54 54 54 54	59 TRAFF I % TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56 79 55 85 60 101 71 111 7	C LOG RELATE AT 100 LBS 97 105 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079 1138	PAVE ANALY FLEX 465 3046 3069 3194 2392 2518	P 4ENT (SIS RIDG 555 3617 3644 3794 2846 2997
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01660 01660 01680 01690	5 52 1969 VEH MIL 110 226 651 830 782 651 581	MAR 1, LUBBOCK .E ADM Y 95 0 9 10 0 9 14 0 9 14 0 9 13 0 9 13 0 9 13 0 9	1971 DESIC (XX.XX) 0 600 0 540 0 510 0 860 0 800 0 1120 0 710	PLANN GN YEAR ESTIM AOT 2530 32660 32590 34510 21550 21860 14660	ATED V.M. 279 49 17	ИЕЧ Г К FAC 10 10 10 10 10 10	DIR DIST 54 54 54 54 54 54 54 54 54	59 TRAFFI % TRUCKS IN IN ADT DHV (XX.X) 162 118 79 56 79 55 85 60 101 71 111 78	C LOG RELATE AT 100 LBS 97 105 105 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079 1138 867	PAVEN ANALY FLEX 465 3046 3069 3194 2392 2518 1919	P 4ENT (SIS 81DG 555 3617 3644 3794 285 2997 2285
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01650 01660 01690 01690 01700	5 52 1969 VEH MIL 110 226 651 830 782 651 830 782 651 830	MAR 1, LUBBOCK LE ADM Y 95 0 9 90 0 9 90 0 9 91 0 9 92 0 9 93 0 9 93 0 9 93 0 9 93 0 9 93 0 9 93 0 9	1971 DESIC (XX.XX) 0 600 0 540 0 540 0 540 0 860 0 800 0 1120 0 710	PLANN GN YEAR ESTIM AOT 2530 32660 32590 34510 21550 21860 14650 10910	ING SURV	ИЕЧ Г К FAC 10 10 10 10 10 10 10	DIR DIST 54 54 54 54 54 54 54 54 54 54 54	59 TRAFFI % TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56 79 55 85 60 101 71 111 78 117 82 138 97	C LOG RELATE 100 LBS 97 105 105 105 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20 20 20 20 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079 1138 867 762	PAVEN ANALN FLEX 465 3046 3069 3194 2392 2518 1919 1682	P 4ENT 7SIS 81DG 555 3617 3644 3794 2846 2997 2285 2005
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01650 01660 01690 01690 01710	5 52 1969 VEH MIL 110 226 651 830 782 651 588 218 52	MAR 1, LUBBOCK .E ADM Y 95 0 9 10 0 9 14 0 9 10 0 9 14 0 9 13 0 9 13 0 9 13 0 9 13 0 9 14 0 9 13 0 9 13 0 9 14 0 9 13 0 9 14 0 9 15 0 9 14 0 9 15 0 9 16 0 9	1971 DESI(R FAC (XX.XX) 0 600 540 0 510 0 540 0 510 0 710 0 710 0 710	PLANN GN YEAR ESTIM AOT 2530 32660 32590 34510 21550 21860 14650 10940	A TED 2.70 4.9 1.7 4.7 4.7 4.7 4.7 4.7 4.7 4.7 4	ИЕЧ Г К FAC 10 10 10 10 10 10 10 10	DIR DIST 54 54 54 54 54 54 54 54 54 54 54 54	59 TRAFFI % TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56 79 55 85 60 101 71 111 78 117 82 138 97	C LOG RELATE 100 LBS 97 105 105 105 105 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20 20 20 20 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079 1138 867 762 764	PAVEN ANAL FLEX 465 3046 3069 3194 2392 2518 1919 1682 1687	P 4ENT (SIS RIDG 555 3617 3644 3794 2846 2997 2285 2005 2010
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01650 01660 01660 01690 01700 01710 01720	5 52 VEH MIL 110 226 651 830 782 651 588 218 52 57	MAR 1, LUBBOCK E ADM Y 95 0 9 10 0 9 14 0 9 13 0 9 13 0 9 13 0 9 13 0 9 13 0 9 13 0 9 14 0 9 15 0 9	1971 DESI((XX.XX) 0 600 0 540 0 510 0 860 0 800 0 1120 0 710 0 710 0 710 0 710	PLANN GN YEAR ESTIM AOT 2530 32660 32590 34510 21550 21860 14650 10910 10940 110	A TED V. M. 270 49 17 17 13 31	ИЕЧ Г К FAC 10 10 10 10 10 10 10 10 10 10 10	DIR DIST 54 54 54 54 54 54 54 54 54 54 54 54 54	59 TRAFFI TRUCKS IN IN ADT DHV (XX.X) 162 118 79 56 79 55 85 60 101 71 111 78 117 82 138 97 138 97	C LOG RELATE AT 100 LBS 97 105 105 105 105 105 105 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20 20 20 20 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079 1138 867 762 764 769	PAVEN ANALX FLEX 465 3046 3069 3194 2392 2518 1919 1682 1687 1698	P 4ENT (SIS RIDG 555 3617 3644 3794 2846 2997 2285 2005 2010 2023
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01650 01660 01680 01690 01700 01710 01720 01730	5 52 1969 VEH MIL 110 226 651 830 782 651 588 218 57 105	MAR 1, LUBBOCK E ADM Y 25 0 9 30 0 9 30 0 9 3 0 9 10 9 10 9 10 9 10 9 10 10 9 10 9 10	1971 DESI((XX.XX) 0 600 540 0 510 0 860 0 800 0 1120 0 710 0 710 0 710 0 710 0 710	PLANN GN YEAR ESTIM ADT 2530 32660 32590 34510 21550 21860 14650 10910 10940 10947	A TED V.M. 279 49 17 17	F AC	DIR DIST 54 54 54 54 54 54 54 54 54 54 54 54 54	59 TRAFFI % TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56 79 55 85 60 101 71 111 78 117 82 138 97 138 97 139 98	C LOG RELATE AT 100 LBS 97 105 105 105 105 105 105 105 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20 20 20 20 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079 1138 867 762 764 769 764	PAVEN ANALX FLEX 465 3046 3069 3194 2392 2518 1919 1682 1687 1698 1698	P 4ENT (SIS RIDG 555 3617 3644 3794 2846 2997 2285 2005 2010 2023 2010
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01650 01660 01660 01680 01690 01700 01710 01720 01730 01740	5 52 VEH M11 110 226 651 830 782 651 588 218 57 10 257	MAR 1, LUBBOCK E ADM Y 25 0 9 30 0 9 3 0 9 10 0 9 10 0 9 10 0 9 10 10 0 9 10 0	1971 FAC (XX.XX) 0 540 540 540 510 860 91120 710 710 710 710 710 710 710 710 710 710 710 710	PLANN GN YEAR ESTIM ADT 2530 32660 32590 34510 21550 21860 14650 10910 10940 10940 10947 107 47 107 47	A TED V.M. 279 49 17 17 17	F AC	DIR DIST 54 54 54 54 54 54 54 54 54 54 54 54 54	59 TRAFFI % TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56 79 55 85 60 101 71 111 78 117 82 138 97 138 97 139 98 141 99	C LOG RELATE AT 100 LBS 97 105 105 105 105 105 105 105 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20 20 20 20 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079 1138 867 762 764 769 764 752	PAVEN ANALX FLEX 465 3046 3069 3194 2392 2518 1919 1682 1687 1698 1698 1698 1696	P 4ENT (SIS RIDG 555 3617 3644 3794 2846 2997 2285 2005 2010 2023 2010 1978
TRICT - NTY - 1 SERIAL NO. 01620 01630 01640 01650 01660 01680 01690 01700 01710 01720 01730 01740 01769	5 52 1969 VEH M11 110 226 651 830 782 651 588 218 57 100 253 215	MAR 1, LUBBOCK E ADM Y 25 0 9 30 0 9 3 0 9 10 0 9 100 0 9 100 0 9 100 0 9 100 0 9 100 0 9 10	1971 R FAC (XX.XX) 600 540 510 860 800 1120 710 710 710 710 710 710 710 710 710 710 710 710 710 710 710	PLANN GN YEAR ESTIM ADT 2530 32660 32590 34510 21550 21860 14650 10910 10940 10950 100500 100500 100500 100500 100500 100500 100500 100500	A TED V.M. 279 49 17 17 17 17 17 17 17 17 17 17 17 17 17	F AC	DIR DIST 54 54 54 54 54 54 54 54 54 54 54 54 54	59 TRAFFI % TRUCKS IN IN ADT DHV { XX.X } 162 118 79 56 79 55 85 60 101 71 111 78 117 82 138 97 138 97 138 97 139 98 141 99 144 101	C LOG RELATE AT 100 LBS 97 105 105 105 105 105 105 105 105 105 105	ED TRAFFI HWLD % TANDEM AXLES 20 20 20 20 20 20 20 20 20 20 20 20 20	C DATA TNWLGT 8000 LBS (K) 211 1369 1379 1437 1079 1138 867 762 764 769 764 752 748	PAVEN ANALN FLEX 465 3046 3069 3194 2392 2518 1919 1682 1687 1698 1698 1698 1696 1650	P 4ENT (SIS RIDG 555 3617 3644 3794 2846 2997 2285 2005 2010 2023 2010 1978 1967

This page replaces an intentionally blank page in the original --- CTR Library Digitization Team

APPENDIX D

ROAD LIFE (RL-1) CODING FORM

This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

file 10	.63-1	11510	1 ALD 1	2047) 11 	icu 7f	AUN	<u>es</u>						CON	STRI	TEXA JCTI(IS H	ECO	NAY DE RD FOR	PARTI ROAU	HENT	E STI	UDY							C	ARD NO.	
ist. A	sphalt:	155.	on Crs.	2 142 175	nd		<u>л/Ж</u> Зт									Th	ia Jo	ob Retire:	9										L		i
182. A	ers.	151.		2	cd		31	1			Contr	·			Sect.			Job		((Card 1	io)	Contro	1 No		Sec.	J	ob	
<u>schalt</u>	: Kin:										Contr	·		.	Sect.			lop		(0	Card I	fo)	Highwa	y No. Sta	ite		U_S		
XISTIN	G WORK:		DESCRIP	TION AND	REMAN	K3					Contr	•			Sect.			Job		((Card !	lo)	Proj.	Nc					
Grad	102	Bas	e	Surf		Sr.1	yp.	Job	Yeal	r	Sta.													_	Dist.	c					
											Sta.														Compil	ed ty			Date		
				·																						- KR - 00-0-	<u>/////////////////////////////////////</u>	<u>f opei</u> d	RATICUS		
w Vork	L				i			L			From:				•									-		CC-L	Code a			コ	
	*										To:															Strip P.C.P	Hap P L. Cole	osted i		-i	
				· · · ·																					51	r. Nort-r	5	30.	01	istal	
	_										Equat	ions																Str	-0	وم المعملية وعور	\neg
											n										-						<u> </u>	1		F	1 .
											Excep	tions													Rd. way	F۱.		•	×	Mile	s
. 1			1.	<u> </u>	<u> </u>	11-	17						1								r		T	-		T		1			_
'	2	,	•	,	6	'	9	•	ю		2			14	15	16		17		18	[19	20	21	22	23 24	21	2	16 27		
								5 .	u u			Date Let	Con	te mpl.	H B D	Ŷ				e		υ			2.	125	3	1	12		
			er	Code	- inde		las	Sys	1123	8					Sy's	4	1	Kind of		606 61 1		ecti	ldrs	haf.	actr		Cole		S Miles		
5	rol	lor	Mum	te	- A	l ŝ	17	A BA	S.L.F.	Sys	ŭ				5	ace		WOLK		01.05 5-53		28-S	á_	- H	0 10 10	3 6 5 5	i k			9	
Coun	Cont	Sect	Tob	TX I	Card	Pars	14BV	ELSP	Tes T	Aid		}			Year	Surf				Croe		Croi			1.0		8.4			SXI	
-	-									-		onth	outh	- Ser									YPe	ype	A60	18 - 18 - 18 - 18 - 18 - 18 - 18 - 18 -					
23	4 5 6	7 6 9	10 11 1	2 13 14 15	16 17	18 15	20 2	1 22 21	3 24 25	26 27 2	8 29 30	2 >	33 34 35	36 37	30 39	40 41	42 43	3 44 45 46	47 48 4	9 50 81	52 53	54 55 56	57 58 59	60 6	62 63 (65 66 67	68 69	10 71 1	72 73 74 75 7	6 77 78 79	80
_				002	2	1						1		1						<u>-</u> ,						7 I T					
		7	1		1	Π	Ţ	28		2	9	Г	30	1	31	32	50		51										1		7
							S	ubgrad	le							2		1													
				2	Ŀ	1	;		1							Ite													F		
	7	Ę	19561	Cod	- deno				•]		ध्रम्	1	Lat		S int	10			ollars			, r	bilars			Do I	laru		0 1: 31	ALL.	
t,	orre	ctio	4 M.	date	20	111		dth 1	bth	¥ile	nt an		gfnu lepc		lepc	aber	e B				E.	-			E				Ge		
ខ	ĉ	ŝ	30	dŋ	5	E .	\$} 2] ;	ŝ	12		Re		a z		₩¥.	Ň	It				H H	1			I A			- I	Ĝ.		
2 3	4 5 8	7 8 2	10 11 1	2 13 14 13	16 17	18 19	20 2	22 2	24 25	78 27 2	8 29 30	31 32	33 34 35	30 37	38 39	40 41	47 4	3 4 4 4 5 4 6	47 48 4	9 50 51	52 53	54 35 56	57 56 59	60 6	1 67 63	64 65 65 67	68 69	70 71	72 73 74 75 7	6 77 78 75	80
-			1	002		2		1				1				1															
			1		1		1	1 -		1	· ·	<u> </u>									1			1							Π,
				ę	1	5	5													1	1					•					
	۲.	g	19 qui	S	- đung	1.40			De 3				Dell	1]	Dollar				Doil=	r.a.			Dollars			Della	178	
ounto	'ntrc	otto	N N	date	Td N	122			101	Lars		8	DOT	101.2		E		юдаr		8		W-1101		H				H			3
ŏ	ပိ	× ا	ء م	5	d,	6	H H					#				11				H				If				#			3
																l									_						\square
23	4 5 8	7 4 9	10 11 1	2 13 14 1	16 17	18 15	20 2	22 2	3 24 25	26 27 2	8 29 30	31 32	33 34 35	36 37	38 39	40 41	42.4	3 44 45 45	47 48 4	2 50 51	52 63	54 55 56	87 88 31	60 6	82 63	84 65 66 6	7 68 69	70 71	72 73 74 75 7	6 77 76 79	80
		1.		002		3																									

This page replaces an intentionally blank page in the original --- CTR Library Digitization Team

ROAD INVENTORY (RI-2) CODING FORM

APPENDIX E

•

This page replaces an intentionally blank page in the original. -- CTR Library Digitization Team

Jol	b Number A.02			fi Mas	le Name RI-2 STER FILE		Storage Medium TAPE				
Avg.	No. Reco	rds	Fi	elds 39	Cha	racters 103	How Current	Date Prepare Dec. 1968			
File	Sequence	1. CO	UNTY, 2. T. 5. ENI	CONTROL DING MIL	L, 3. SEC	TION ·	Prepared By B.F.B.	Reviewed By			
Label	s LRE	CL=103,	BIKSIZE=]	.030				Record Sourc			
Reten Remar	tion Cha ks	racteri	stics					Pagé 1 of			
Item No.	From . To	Size	No. of Dec. Pos.	Field Char.	Label		Item Name				
1	1-2	2		I	DIST	DISTRICT NUMBE	R ·				
2	3-5	- 3		I	CONTY	COUNTY NUMBER					
3	6-9	4		I	CONTL	CONTROL					
4	10-11	2		I	SHD	SHD SECTION NU	MBER	-			
5	12-16	5	3	I	ISTART	BEGINNING MILE	POINT				
6	17-21	5	3	I	IEND	ENDING MILEPOI	NI				
7	22	1	•	I	ISTAT	HIGIWAY STATUS					
8	23-24	2		I	ADMI2	AMINISTRATIVE	SYSTEM				
9	25-26	2		A	FASY2	FEDERAL AID SY	STEM				
10	27-28	2		A	INFE2	IN FEDERAL RES	ERVATION				
11	29-33	5		I	ICTY	CITY NUMBER					
12	34-37	4		А	· LCC24	RIGHT OF WAY W	IDIH	MAIN LANES			
13	38-40	3		A	10027	ROAD BED WIDTH		MAIN LANES			
14	41-43	3		A	SURFWD	SURFACE WIDTH		MAIN LANES			
15	44 ·	1		A	L0C31	BASE TYPE		MAIN LANES			
16	45-46	2		A	10032	SHOULDERS TYPE		MAIN LANES			
17	47-48	2		I	SURFIP	SURFACE TYPE		MAIN LANES			
18	49	1		MAIN LANES							
19	50	1		I	DESIGN	DESIGN TYPE	•	MAIN LANES			
						1	anna baile 's antif Paillithanag				
20	51-52	2		A	K0C37	NUMBER LANES		MAIN LANES			

.

•

•

Joi	b Number			Fi MAS	le Name RI-2 IER FILE		Storage Medium TAPE				
Avg.	No. Reco	rds	Fi	elds	Cha	racters	How Current	Date Prepared			
File :	Sequence						Prepared By B.F.B	Reviewed By			
Labe 1	5							Record Source			
Reten	tion Cha	racteri	stics					Pagé 2 of 2			
Remar	ks		•				····				
Item No.	From To	Size	No. of Dec. Pos.	Field Char.	Label						
22	5455	2		A		TYPE	FRO	NTAGE ROADS [2]			
23	56-58	3		A		WIDTH	FRC	INTAGE ROADS			
24	59-60	2		A		NUMBER LANE	s <u>fr</u> a	NTAGE ROADS			
25	61	1		A		NUMBER ROAL	<u>s </u>	NTAGE ROADS			
26	62-63	2		A	LOC56	GRADES, NUN	BER, 3-5%				
27	64-65	2		A	LOVER3	GRADES, LEN	IGIH, 3-5%	•			
28	66-67	2		<u>A</u>	LOC60	GRADES, NUN	BER OVER 5%				
29	68-69	2		A	LOVER5	GRADES, LEN	IGTH OVER 58				
30	70-71	2		A	LOC64	CURVES, NUM	BER				
31	72-73	2	l I	A	LCURVE	CURVES, LEN	IGIH				
32	74-75	2		A	SYST2	HIGHWAY SYS	SIEM	······			
33	76-79	4		A	·HWY	HIGHWAY NUN	BER				
34	80	1		I I	ISR	ROUTE SERLA	AL NUMBER	[2]			
35	81	1		I	IDR	DIRECTION	OF ADJUSTMENT				
36	82-86	5	3	I	IADJ	AMOUNT OF A	DUSTMENT	[2]			
3/	87-92	6	3	<u> </u>	NSTART	ADJUSTED BE	GINNING MILEPOIN	<u>r [2]</u>			
<u></u>	93-98	6 	3	I T	NEND	ADJUSTED EN	DING MILEPOINT	[2]			
	99-103	5	5	1	LENGIH	LENGTH OF S	SECTION				
[2]	NOT IN	RECORD	AT THIS	IME.							

.

: : :

THE AUTHORS

Oren G. Strom, an active duty U. S. Air Force Civil Engineer, has recently completed a graduate engineering research program at The University of Texas at Austin. His experience includes 18 years of military engineering management, both in the U. S. and overseas, and one year of civilian employment with the Wisconsin State Highway Commission.

W. Ronald Hudson is an Associate Professor of Civil Engineering at The University of Texas at Austin. He has had a wide variety of experience as a research engineer with the Texas Highway Department and the Center for Highway Research at The University of Texas at Austin and was Assistant Chief of the Rígid Pavement Research Branch of



the AASHO Road Test. He is the author of numerous publications and was the recipient of the 1967 ASCE J. James R. Croes Medal. He is presently concerned with research in the areas of (1) analysis and design of pavement management systems, (2) measurement of pavement roughness performance, (3) slab analysis and design, and (4) tensile strength of stabilized subbase materials.

James L. Brown is a Senior Designing Engineer in the Pavement Design Section of the Texas Highway Department, Highway Design Division. He has a variety of experience at both the district and division levels and has been actively involved in the development and pilot implementation of the Flexible Pavement System since its conception. He is the author of several publications in highway engineering research and is Area Coordinator for the Pavement Research Advisory Committee.

143